The impact of bank regulation on the cost of credit: Evidence from a discontinuity in capital requirements

Emilia Bonaccorsi di Patti, Mirko Moscatelli and Stefano Pietrosanti^{*}

April 24, 2023

Abstract. We study the effect on credit relationships of the Small and Medium Enterprises Supporting Factor (SME-SF), a regulatory risk weight reduction on small loans to SMEs. Employing a regression discontinuity design and matched bank-firm data from Italy, we find that a 1 percent drop in capital requirements causes an average 13 basis points reduction in the cost of credit. Moreover, with a novel measure of bank regulatory capital scarcity, we show that the drop is larger for banks facing tighter constraints. Furthermore, the drop is larger for firms with low switching costs, while the sharp assignment rule may have led to the rationing of marginal borrowers. Such findings indicate that the entire distribution of firms and banks' characteristics plays a crucial role in determining the impact of regulatory capital changes.

Keywords: Capital requirements, SME, Cost of credit, Credit access, Switching costs.

JEL Classification: E51, G21, G28.

^{*} Emilia Bonaccorsi di Patti: Bank of Italy, emilia.bonaccorsidipatti@bancaditalia.it; Mirko Moscatelli: Bank of Italy, mirko.moscatelli@bancaditalia.it; Stefano Pietrosanti, Bank of Italy, stefano.pietrosanti@bancaditalia.it. We thank Mitchell Berlin, Marcello Bofondi, Nicola Branzoli, Pablo D'Erasmo, Roberto Felici, Giorgio Gobbi, Edison Yu, Francesco Palazzo, Diane Pierret (discussant), Michael Schwert, Enrico Sette, Holger Sieg, André F. Silva, Rhiannon Sowerbutts (discussant), four anonymous referees and seminar participants at the Bank of Italy, University of Pennsylvania, Wharton, Bank of Denmark, Federal Reserve Board, Bank of England, LUISS, Università di Bologna, Federal Reserve Bank of Boston, 4th ECB Macroprudential Policy and Research Conference and the 6th ESCB Research Cluster 3 Workshop. The authors wish to express their special thanks to Francesco Manaresi, for insightful comments and suggestions. Stefano Pietrosanti wishes to express his special thanks to Guillermo Ordóñez, Michael Roberts, Alessandro Dovis, Marc Flandreau and Petra Todd for providing invaluable guidance throughout the preparation of this work. The opinions expressed do not necessarily reflect those of the Bank of Italy, the Eurosystem or their staffs. All errors are our own.

I Introduction

Bank regulators have employed minimum capital requirements to ensure bank solvency since the introduction of the Basel Accord framework in the 1980s. More recently, minimum capital requirements have become part of the macroprudential policy toolkit, which includes countercyclical changes in mandatory capital buffers to moderate lending booms in good times and mitigate lending busts in bad times (Claessens, 2015).

Minimum capital requirements aim to bring bank leverage closer to the socially optimal level. Banks may engage in excessive leverage because of moral hazard, either induced by limited liability and managerial discretion (Jensen and Meckling, 1976; Myers and Majluf, 1984), or by the distorted incentives arising from deposit insurance and the implicit or explicit government safety net. Imposing minimum capital requirements increases shareholders' stake, thereby reducing the ex ante incentive to gamble with insured deposits (Kareken and Wallace, 1978; Keeley, 1990).

If capital and debt are not perfect substitutes, capital requirements may come at a cost. If bank capital is more costly than debt (Diamond and Rajan, 2000), imposing minimum capital requirements may result in higher interest rates and reduced credit supply. Even though there have been many attempts at assessing the magnitude of such costs, for example through model-based simulations (e.g. Kashyap, Stein, and Hanson, 2010; Miles, Yang, and Marcheggiano, 2012) and through investigation of the effect of negative shocks to banks' capital (e.g. Berger and Udell, 1994; Peek and Rosengren, 2000; Behn, Haselmann, and Wachtel, 2016), little consensus has emerged.

We provide direct evidence on such cost in a quasi-experimental setting, exploiting the Small and Medium Enterprises Supporting Factor (SME-SF), a regulatory capital discount targeting small loans, aimed to shield European SMEs from the adverse effects of the tougher regulation coming with Basel III. From the impact of such discount on the spread on SMEs revolving credit facilities we infer that a 1 percentage point decrease in minimum capital requirements causes a 13 basis points drop in the cost of bank credit.

We exploit the unique framework, the availability of a rich set of firm-level proxies of switching costs, and a novel measure of banks' capital scarcity to gauge the extent to which the pass-through is heterogeneous across firms and banks. First, we show that: (i) borrowers with multiple healthy credit relationships drive the average effect; (ii) better borrowers (high EBITDA) get about a 40 percent larger discount; (iii) worse borrowers (high drawn over granted credit ratio or high risk score) get a 50 to 60 percent smaller discount. Second, banks whose shadow cost of capital is higher apply a 20 percent larger than average discount to their borrowers. Third, while there is no impact on credit quantities on average, credit to weaker borrowers grows less after the introduction of the SME-SF. Such evidence suggests that the sharp assignment cut-off could engender undesirable effects. Overall, we highlight that the effects of changes in capital requirements on credit depend on the entire distribution of borrowers' and lenders' characteristics.

We base our analysis on a rich dataset on bank loans to firms from the Italian Credit Register, matched with firm and bank characteristics and covering the period around the introduction of the SME-SF. The SME-SF entered into force on the 1st of January, 2014 through Article 501(1) of the Capital Requirements Regulation (CRR) and reduced risk weights of eligible exposures by 23.81 percent. Considering a corporate loan to an SME with a risk weight of 100 percent and a minimum capital requirement of eight percent of risk weighted assets, the reduction in the minimum capital requirement is approximately two percent.

Under the assumption that potential confounding factors do not change discontinuously at the \in 1.5 million threshold, we employ the SME-SF eligibility rule to estimate the effect of capital requirements on lending rates with a Regression Discontinuity Design (RDD).¹ Through the RDD, we compare credit relationships that are very similar before the reform, but face different risk weights once the SME-SF is implemented. To support the validity of such a design, we provide evidence that firms', banks' and relationships' characteristics do not vary discontinuously at the SME-SF threshold, and that there is no bunching of credit relationships below the threshold in the two years before and the year after policy implementation. The absence of manipulation is not surprising as the estimated drop in the interest rates amounts to a saving in the annual cost of credit of a few thousand Euros for credit lines that are drawn for more than one million Euro. Moreover, we run placebo tests for non-SME and SME relationships before the SME-SF

¹ The approach, introduced by Thistlethwaite and Campbell (1960), is commonly both in labor economics and in empirical corporate finance. For a few example in the last field, see Chava and Roberts (2008), Keys et al. (2010), Agarwal et al. (2017), Rodano, Serrano-Velarde, and Tarantino (2018) and Becker, Opp, and Saidi (2021).

and find no evidence of spurious effects.²

Our baseline analysis shows an average 13 basis points decrease in the cost of credit for a 1 percentage point decrease in capital requirements. This average effect may only partially reflect the benefit accruing to banks from the marginal capital requirements relaxation for several reasons. First, if borrowers face significant frictions to switch between lenders, banks can exercise monopoly power and retain a fraction of the surplus. The change in the cost of credit will then reflect only such fraction, which in turn depends on the size of switching costs.³ Second, the SME-SF may be more beneficial to the borrowers of banks whose regulatory capital is relatively scarcer. In the absence of the SME-SF these banks would raise the cost of credit more than capital-abundant banks as a result of the tighter Basel III regulation.

We exploit our rich dataset to analyze the relevance of these two heterogeneity dimensions. To test for the role of switching costs in influencing the degree of the pass-through across firms we employ different literature-based proxies. To measure the regulatory capital scarcity we instead employ a novel variable based on unique supervisory information at the bank level. When Basel III became effective, banks were given some time to adjust to the more restrictive capital definition. During the phase-in period they had to report to the supervisor what their capital would have been if the new definition were to be applied at once. We employ the difference between such fully phased-in capital and the transitory capital as our measure of relative scarcity. The intuition is that the larger the difference, the more significant an adjustment the bank had to do to revert to the desired buffer level by the end of the phase-in period (Repullo and Suarez, 2013).

Under the assumption that banks transfer the entire benefit of the capital discount to borrowers that enjoy low switching costs, we find that banks may be happy to pay up to 16 cents for each euroof regulatory capital saved. Moreover, we observe that banks

 $^{^2}$ As regards manipulation, we must also bear in mind that firm can only manipulate by not using credit. If they need credit, such a manipulation is costly.

³ The importance of switching costs for the dynamic of credit and its cost has been documented, for example, by Ioannidou and Ongena (2010); Barone, Felici, and Pagnini (2011); Allen, Clark, and Houde (2019). For theoretical works on the effect of banks monopoly power on the cost of credit, we refer instead to Sharpe (1990) and Petersen and Rajan (1995). Moreover, as low switching costs should be an important driver of the pass-through, a fixed effect identification strategy would result in a larger effect due to sample selection. The pass-through would be identified only for firms with multiple relationships. These firms are likely to be less captive than single-bank borrowers and would therefore receive greater discounts (Ioannidou and Ongena, 2010). In the Appendix we demonstrate that this intuition holds true, documenting an increase in the point estimate when we restrict the estimation to the sub-sample of firm with multiple bank relationships and we show that such increase can be fully attributed to sample selection.

tend to increase granted credit less on those eligible lines whose utilization is closer to the maximum granted amount, although we do not find any significant average effect on the amount of credit granted. This result suggests that risk-weight rules incentivizing credit provision may have unintended effects on some groups of borrowers when based on sharp cut-offs. From a policy perspective, the subsequent decision of substituting the SME-SF cut-off with a smoother tapering of the discount was appropriate.

Related literature: This paper contributes to the literature on the impact of minimum capital requirements on the supply of credit to firms (Aiyar, Calomiris, and Wieladek, 2016; Behn, Haselmann, and Wachtel, 2016; Jiménez et al., 2017; Mayordomo and Rodríguez-Moreno, 2018) and the one trying to quantify the costs of capital regulation for banks (e.g. Kashyap, Stein, and Hanson, 2010; Miles, Yang, and Marcheggiano, 2012; Kisin and Manela, 2016; Plosser and Santos, 2018; Glancy and Kurtzman, 2022), providing a novel view into the distributive effects of capital regulation at the relationship-level.

Our assessment of the average pass-through is considerably larger than the one suggested by Kisin and Manela (2016), who derive the shadow cost of capital requirements from the extent to which banks exploit a costly loophole in regulation.⁴ Our estimates are also larger than those presented in the two quasi-experimental studies by Plosser and Santos (2018) and Glancy and Kurtzman (2022). We suggest that this difference is due to the context of our study. In contrast with Plosser and Santos (2018), which exploits variation from Basel I and II, we focus on the European implementation of Basel III. Basel III introduced a more significant tightening of capital regulation, which may have had a non-linear effect on the cost of regulatory capital for banks. Regarding Glancy and Kurtzman (2022), we focus on the universe of banks operating in Italy, which may face higher funding costs than the large US banks on which they focus instead.

Uniquely, we highlight how the effect of capital regulation on the cost of credit depends on the entire distribution of firms' and banks' characteristics, lending support to the conclusions of recent theoretical works such as Ambrocio and Jokivuolle (2017); Bahaj and Malherbe (2020); Harris, Opp, and Opp (2020). We show evidence of how firms' cost of switching between lenders can influence the pass-through, suggesting a significant

⁴ For a more in depth discussion of the modeling assumptions that are important to explain the small estimates by Kisin and Manela (2016) we refer to Plosser and Santos (2018)'s introduction. In brief, Kisin and Manela (2016)'s calculation assumes that banks can move on-balance sheet assets freely and at a low cost assets to off-balance sheet conduits; relaxation of such hypothesis may reconcile the discrepancy between our findings and theirs.

and under-explored link between the capital requirements literature and the literature on the effects of monopoly power within the context of credit relationships (for the latter, see Santos and Winton, 2008, 2019).⁵ Furthermore, we document that banks expecting a greater drop in regulatory capital from the Basel III reform decrease rates more, showing a direct link between the heterogeneity in the shadow cost of capital and the pass-through.

Our approach to measuring the shadow cost, similarly to the one in Plosser and Santos (2018), does not rely on a difference-in-difference plus fixed effects strategy. The latter approach has been recently subject to methodological revision (see, e.g. De Chaisemartin and d'Haultfoeuille, 2020) and has been called into question in its corporate finance applications (Berg, Reisinger, and Streitz, 2021; Paravisini, Rappoport, and Schnabl, 2022).⁶

Finally, our results shed light on the effectiveness of risk weights as a policy instrument. Targeted changes in risk weights are increasingly being employed as a macro-prudential policy instrument(see, e.g. Altunbas, Binici, and Gambacorta, 2018). We add to the growing literature on the effects of such policies, e.g. Akinci and Olmstead-Rumsey (2018), Mayordomo and Rodríguez-Moreno (2018) and Lecarpentier et al. (2020) for the SME-SF in particular. While Mayordomo and Rodríguez-Moreno (2018) and Lecarpentier et al. (2020) study the effect of the SME-SF on credit access, we complement their analysis by studying the impact on the cost of credit. We also show evidence suggestive of an unintended effect, because exposures close to the eligibility threshold to firms with higher switching costs grow less than others. In this sense, our findings are in line with those of Becker, Opp, and Saidi (2021), who show that sharp changes in risk weights buckets applied to insurance companies' assets lead to strategic manipulation by intermediaries.

The paper proceeds as follows: Section II provides background information on Basel III, with special focus on the SME-SF and the transitory measure relaxing the novel capital standards; Section III describes our data; Section IV explains our identification

⁵ As regards the effects of capital regulation, the only important exception we are aware of is Corbae and D'Erasmo (2021), which uses a large, general equilibrium model of dynamic monopolistic competition between lenders to track the effects of regulation on lending concentration and ultimately on the cost and availability of credit. A growing literature is instead tackling the importance of banks' monopoly power for the transmission of monetary policy, highlighting similar results (see, e.g. Agarwal et al., 2023; Drechsler, Savov, and Schnabl, 2017; Wang et al., 2020; Benetton and Fantino, 2021).

 $^{^{6}}$ Other empirical banking studies employing RDD techniques are Rodano, Serrano-Velarde, and Tarantino (2018), which studies access to credit over the cycle through a firm-level discontinuity in the assignment of credit ratings, and Becker, Opp, and Saidi (2021), which focuses on insurer's balance sheets instead of corporate loans and exploits risk-weight discontinuities at the instrument level.

strategy; Section V illustrates and interprets the results; Section VI concludes.

II Institutional Background

The three key elements of capital requirements are: Minimum regulatory capital ratios, risk weights for each asset or asset class, and rules defining what counts as capital from a prudential perspective. After the Global Financial Crisis, the Basel Committee on Banking Supervision approved new capital standards (Basel III) with the purpose of increasing the quantity and quality of the capital buffer that banks need to hold against their risk weighted assets. The European Union adopted the new standards in June 2013, with application starting on January 1, 2014;⁷ some of the measures entered into force immediately while others gradually.⁸

The framework put forth by the Basel Committee requires banks to hold at least 4.5 percent of risk weighted assets in Common Equity Tier 1 (CET1),⁹ and increases the minimum Tier 1 capital requirement from 4 to 6 percent while leaving the overall requirement at 8 percent. Under Basel III banks are also required to hold two additional buffers: the Capital Conservation Buffer and the Countercyclical Capital Buffer. The first consists of an additional CET1 buffer of 2.5 percent of risk weighted assets; the second is a CET1 buffer that varies between 0 to 2.5 percent of risk weighted assets depending on cyclical conditions in the credit market.¹⁰ Finally, the new rules tightened the capital definitions, to grant uniform, high quality buffers for loss absorption.

Considering that under the previous framework (Basel II) banks were required to hold

 $^{^7}$ See the European Commission's Online References at https://ec.Europa.eu/info/law/banking-prudential-requirements-directive-2013-36-eu_en.

⁸ and its implementation, On Basel III see the Basel Committee's "Basel III: global regulatory framework for more resilient banks and banking systems" atA https://www.bis.org/publ/bcbs189.pdf, and their updated summary in "High-level summary of Basel III reforms" at https://www.bis.org/bcbs/publ/d424_hlsummary.pdf.

⁹ The definition of CET1 includes "Common shares issued by the bank that meet the criteria for classification as common shares for regulatory purposes (or the equivalent for non-joint stock companies); Stock surplus (share premium) resulting from the issue of instruments included CET1; Retained earnings; Accumulated other comprehensive income and other disclosed reserves; Common shares issued by consolidated subsidiaries of the bank and held by third parties (ie minority interest) that meet the criteria for inclusion in CET1 capital [...] and Regulatory adjustments applied in the calculation of CET1" (Basel Committee, 2011, Global Regulatory Framework Report's p.13). Additional Tier1 includes other types of shares; Tier2 capital includes some subordinated debt instruments.

¹⁰ These figures are the fully phased-in buffers; the time-line of implementation is described in the "Basel III phase-in arrangements" document by the Basel Committee at https://www.bis.org/bcbs/basel3/basel3_phase_in_arrangements.pdf. We will provide more details regarding the transition to the new regulatory regime in the last part of this Section.

an overall 8 percent capital buffer, while under the new fully phased-in rules the buffer would be at least 10.5 percent, European banks and other stakeholders raised the concern that the reform would lead to an excessive tightening of the credit supply, particularly to SMEs, hampering the recovery of the EU economy.¹¹

In response to this concern, the EU capital regulation adopting Basel III in the EU (Capital Requirements Regulation - Capital Requirements Directive IV, CRR-CRD IV henceforth) introduced measures to smooth the transition. We will make avail of two such measures in this study: The Small and Medium Enterprise Supporting Factor (SME-SF), which will help us identify the effects of capital regulation changes; the transitory regime for the adoption of the more stringent definition of capital, which will help us investigate bank-level heterogeneity.

II.1 The SME Supporting Factor

The SME-SF is a 23.81 percent discount on the risk weight that applies to loans granted to firms with turnover below \in 50 million, provided that the total exposure of the lender to each eligible firm is below \in 1.5 million. The magnitude of the SME-SF exactly counteracted the maximum overall increase in capital requirements implied by the additional Capital Conservation Buffer.¹²

The Capital Conservation Buffer was gradually phased in between 2016 and 2019, but the SME-SF became effective on January 1, 2014. Consequently, capital requirements for outstanding and new eligible exposures to SMEs were de facto lowered with respect to the pre-CRR/CRD IV framework. To give an example of the SME-SF effect on minimum capital requirements, we consider an average capital requirement of 8 percent and a pre-SME-SF risk weight of 100 percent. After the implementation of the SME-SF, the minimum capital requirement on an SME's credit line utilized for \leq 1.6 million would be unchanged at \leq 128,000. Instead, the minimum requirement on a \leq 1.4 million SME exposure would amount to \leq 85,000, taking the SME-SF into account. Such stark change in minimum capital requirements at the SME-SF eligibility threshold provides ground to

¹¹ For a more detailed comparison between the Basel II and Basel III regimes, we refer to Gatzert and Wesker (2012). Regarding the concern of European stakeholders about the strictness of Basel III's rules, see Recital 44 of the "Regulation (EU) No 575/2013 of the European Parliament and of the Council" available at https://eur-lex.Europa.eu/legal-content/EN/TXT/?uri=celex%3A32013R0575.

 $^{^{12}}$ A 23.81 percent reduction in a pre-reform risk weighted exposure of 100 would exactly compensate for the increase in the capital ratio: from the 0.08*100 implied by Basel II, to the equivalent 0.105*76.19 under the fully phased-in Basel III regime.

expect an effect on loan pricing.

Anecdotal evidence suggests that the SME-SF did influence credit supply for targeted SMEs. According to the Intesa San Paolo Bank¹³ response to the Call for Evidence on the SME-SF by the European Banking Authority (EBA):

Despite being difficult to quantify the exact price reduction triggered by the application of the SMEs supporting factor, a direct relation between the SMEs SF and the credit price is easy to draw as the cost of regulatory capital is one of the key components of the credit pricing models. The possibility of applying the SF on the eligible SMEs exposures significantly reduces the cost of regulatory capital for such exposures; this capital relief ensures a direct (positive) effect of the SF on the credit price for SMEs borrowers.

In the same vein, the German Banking Industry Committee responded that:

The SMEs Supporting Factor reduces own funds requirements and cuts the cost of capital. This is all the more important the higher interest rates climb, because customer price sensitivity then also increases. If interest rates are expected to rise, cost of capital is thus likely to become more important [...] A lower cost of capital increases profit margins and makes SME loans more attractive.

Even so, the initial effort by the EBA (EBA, 2016) to evaluate the effect of the SME-SF on lending has returned no strong evidence in favor of an immediate effect. However, the EBA's analysis is based on survey data and, for this reason, it cannot fully disentangle supply from demand, or account for the confounding effects of other aspects of Basel III implementation in Europe.

Two recent studies tackled such identification problems using micro-data, and both found evidence of a positive effect of the SME-SF on lending. The first, Mayordomo and Rodríguez-Moreno (2018), finds that the SME-SF contributes to easing credit constraints of medium-sized firms. The second, Lecarpentier et al. (2020), finds instead a lagged, positive overall effect on credit supply, stronger for very small loans of small and micro firms. As both these works find evidence of an effect on credit supply conditions, we argue

 $^{^{13}}$ One of the largest Italian banking groups.

that the SME-SF provides a promising testing ground to improve our understanding of the effects of minimum capital requirements regulation. In particular, the effect of the SME-SF on loan rates is still not explored. Our objective in this paper is the investigation of this aspect, which gives us a chance to learn more about the broader issue of the cost of capital requirements to banks.

II.2 The New Capital Ratio and the Transitory Regime

The Basel III reform aimed to increase the quality of capital by tightening the definition of the highest quality capital, i.e. CET1. First, CET1 is distinguished from additional Tier 1 capital, the latter being constituted by all unsecured and perpetual non-common share instruments. Second, Under Basel III certain items must be deducted from CET1 capital (intangibles, deferred tax assets, gains from securitization transactions, cross-holdings, and investments in the capital of financial institutions out of the scope of regulatory consolidation). Finally, the new regime requires banks to hold greater Tier 1 buffers against third parties' equity and securitization exposures.¹⁴

An immediate application of the new definition of high quality capital would have been tough on banks. For example, the EBA (2014) Basel III monitoring exercise reports that the CET1 ratio of large banks (Tier 1 capital greater than \in 3 billion and internationally active) would have dropped from 11.9 to 9.1 percent if the new rules were applied altogether. For all the other banks the CET1 ratio would have dropped even more, from 12.4 to 8.8 percent.

To avoid an abrupt drop in the capital ratio's numerator, the Basel Committee (2011) Global Regulatory Framework Report recommended a gradual phase-in of the new capital definition (Section C, paragraphs (c) and beyond).¹⁵ The implementation of the phase-in was left to national regulators. The European Union's CRR Article 478¹⁶ established the deadlines for the implementation of the transitory framework. The Bank

¹⁴ Under Basel II, these exposures qualified for either a 50 percent deduction from Tier 1 and 2 capital or favorable risk weighting. Under Basel III, they require instead further buffer accumulation and are risk-weighted at 1,250 percent. For the full Basel III definition of Tier 1 capital, we refer to Basel Committee (2011, p.15-16) and to Basel Committee (2011, p.21-27) for the complete list of mandated deductions.

 $^{^{15}}$ Paragraph (d), p.28, suggests a broad time frame for adoption, asking banks to comply with "0% of the required deductions on 1 January 2014, 40% on 1 January 2015, 60% on 1 January 2016, 80% on 1 January 2017, and reach 100% on 1 January 2018".

¹⁶ Available at the European Banking Authority's https://www.eba.Europa.eu/regulation-and-policy/single-rulebook/interactive-single-rulebook/1072.

of Italy's instructions reflected the guidelines and broadly matched the Basel Committee suggestions.¹⁷ We refer to capital ratios computed under these transitory rules as the "transitory" capital ratios.

The Bank of Italy's supervisory reports contain detailed information on the transitory and the fully phased-in capital ratios for Italian banks. We employ this information and measure the hypothetical drop in regulatory capital each bank would face in the absence of the transitory regime as the difference between the two. We provide details on this variable's construction in the next section. We interpret it as the "distance" that each bank would have to go to meet the new minimum requirement and on top of that restore its desired capital buffer.

III Data and Measurement

We construct our dataset by matching information on loan quantities and interest rates from the Italian Credit Register and the Bank of Italy archive on interest rates (TAXIA). In addition, we source balance sheet data on borrowers from the provider Cerved and balance sheet information on lenders from the Supervisory Files on banks and banking groups.

The Italian Credit Register contains detailed monthly information on all loans issued by banks and other credit intermediaries above the minimum threshold of \in 30,000, irrespective of whether disbursed or not. TAXIA includes information on interest rates on loans to borrowers that have at least \in 75,000 overall granted or disbursed credit, reported by all but the smallest banks. The TAXIA sample is highly representative as the aggregate value of loans of reporting banks is about 80 percent of outstanding credit. Interest rates are the actual rates paid by each borrower on disbursed credit net of commission and fees. Finally, Cerved is a proprietary database containing firms' balance sheet information and a credit score; total credit to Cerved firms covers about three fourths of loans by Italian banks to the nonfinancial corporate sector.

We obtain such information for the years 2013 - 2014 to investigate the impact of the reform and for years 2012 - 2013 to run placebo tests. We focus on revolving credit

¹⁷ Circular 285 of December 2013 available at the Bank of Italy's https://www.bancaditalia.it/compiti/vigilanza/normativa/archivio-norme/circolari/c285/Circ_285_pub.pdf.

lines' interest rates because, in Italy, these loans are relatively standardized and not collateralized. Moreover, banks can modify the rate on these loans on short notice. Finally, we adjusted our dataset for bank mergers by applying the group structure of 2014 to 2013 and 2013 to 2012. We focus on the top-tier bank holdings for i) capital requirements concern the consolidated entity and ii) group exposure defines SME-SF eligibility.

Our measure of the change in the cost of credit between the pre and the post-SME-SF introduction is the difference between the average rate paid in 2014 and 2013 - winsorized at the upper and lower 2.5 percentile to mitigate the effect of outliers. In addition, we consider a yearly time window as we do not observe when credit lines are re-bargained but only the resulting change in the rates paid. Hence, we want to encompass a period that is long enough to include changes in the cost of the credit line and short enough that it can be reasonable to attribute changes to the implementation of the SME-SF.

III.1 Defining Eligibility for the SME-SF

The SME-SF applied to exposures below $\in 1.5$ million towards firms with gross sales below $\in 50$ million, excluding any amount collateralized by residential real estate.¹⁸ First, we identify eligible firms employing the data on gross sales from the Cerved database.¹⁹ In a given year, we assess firm size using gross sales in the previous year, which is the latest figure that banks can observe as the current balance sheet will be released several months after the closure of the fiscal year.

We then employ the Credit Register data to identify SMEs' credit relationships below the exposure threshold. Eligible relationships are those for which the total credit disbursed is below 1.5 million, regardless of the amount granted. We assess eligibility at the end of period t - 1 when analyzing the change in loan rates from t - 1 to t. For example, we assess the total exposure of credit relationships as of December 31, 2013, while in the placebo tests as of December 31, 2012. The eligibility status is thus a proxy for being "treated" with the SME-SF. First, we notice that this is the best we can do,

¹⁸ For example, if a bank grants a \in 5 million loan and the firm posts residential real estate collateral covering \in 4.2 million, the risk weight discount would apply because the exposure net of the collateral is below the threshold.

¹⁹ This criterion is only one of the three that the European Commission follows to define an SME in other contexts; the other two are that an SME must employ less than 250 employees and hold less than \notin 43 million in assets (see the EU recommendation 2003/361 by the European Commission, available at https://eur-lex.Europa.eu/legal-content/EN/TXT/?uri=CELEX:32003H0361).

as banks do not report treatment status of each credit relationship. Moreover, as long as the correlation between this proxy of treatment assignment and actual treatment assignment is positive and large enough, the effect of mismeasurement will be the attenuation of our estimates. As credit utilization is sticky, and we estimate that lowering capital requirements lowers the cost of treated credit relationships, the above assumption is the most credible.²⁰

We take a number of steps to limit the scope of the mismeasurement concern. First of all, we rely on the fact that, according to the regulation, each bank has to verify the eligibility status of its borrowers and report the amount of SME-SF eligible loans to the supervisors on a quarterly basis.²¹ This implies that at the end of the first quarter of 2014 banks without policies to track SME-SF eligibility can be distinguished from banks with such policies. Using this information, we drop banks that do not report any SME-SF exposures.

On the firm side, we drop relationships involving firms whose SME status appears uncertain – which is, firms with either low revenues but assigned to the size-class "large" or vice-versa; firms assigned to the size-class "large" which report assets below the 43 million \in threshold employed by the EU to define non-SME firms; SMEs that appear to hold a very large number of relationships (more than 11, the last percentile of the number of relationships' distribution).²² Furthermore, we restrict our attention to goodstanding relationships, as the SME-SF only applies to performing borrowers. We also drop collateralized relationships because we cannot distinguish between residential and commercial real estate collateral.²³ Finally, we exclude firms with deeply negative (< -20

 $^{^{20}}$ Intuitively, if lowering capital requirements leads to a decrease in rates and our proxy of treatment is extremely bad, we could estimate a significant *increase* in the cost of credit for what we consider eligible relationships. For an extreme example of how this could happen, we can think of the case in which all observations below the threshold at the end of December 2013 are not assigned to the SME-SF and vice-versa. As we estimate a significant *drop* in rates, we find at worst a lower bound for the actual effect on rates of lowering capital requirements. There is comparatively little work on measurement error in RDD settings when no additional information regarding treatment status is present. One recent paper systematically addressing the topic is Indarte (2023), to which we refer for a deeper discussion.

²¹ The more detailed account we could find about the assessment of eligibility is in the answer to question 2013_417, submitted by an undisclosed bank to the EBA, available at https://eba.Europa.eu/single-rule-book-qa/-/qna/view/publicId/2013_417. The EBA indicates that the requirement must be fulfilled on an ongoing basis, though, the reporting constraint implies that banks must "report to competent authorities every three months their total SME exposures, on the basis of adequate current information".

²² For example, these may be branches of larger firms and conduits for their parent companies' credit access.

 $^{^{23}{\}rm The}$ selection concerns due to this restriction are limited as we focus on revolving loans, which are usually not collateralized.

%) or extremely large EBITDA (> 200 %) or with extremely high (>200 %) or negative leverage. Indeed, extremely high EBITDA may imply deficient assets, while negative leverage indicates extremely negative equity (as we define leverage as debt over debt plus equity). Such firms are likely to either have misreported balance sheet figures – thus, we cannot be sure about their SME status – or are close to default and would add noise to the treatment assignment. Again, banks cannot apply the capital requirement discount to nonperforming borrowers.

III.2 Explanatory Variables

Our dataset includes information on relationships, borrowers, and banks' characteristics that could influence loan interest rates. We use these variables for three purposes. The first is to verify that there are no discontinuous changes in observable characteristics at the SME-SF eligibility threshold; the second, which we discuss in the Appendix, is to increase the precision of our estimate of the impact of the SME-SF (Angrist and Rokkanen, 2015). Finally, we employ some of these variables to explore the heterogeneity in the SME-SF pass-through across firms or banks.

The first set of variables captures the nature of the relationship between the firm and the bank. It includes the lagged ratio of credit disbursed by bank b to firm f to total credit utilized by firm f, which proxies for the importance of the bf relationship to f; the lagged ratio of loans utilized to loans granted for each bf relationship, which proxies for the amount of slack that f has in the relationship with b; the lagged ratio of revolving credit granted to total credit granted on each bf relationship, which captures the intensity of the relationship, as revolving credit lines generate soft information on the firm (Berlin and Mester, 1999).

Moreover, we include a proxy for the distance between the bank and the firm, using a dummy indicating whether the firm and bank locate their headquarters in the same province. The literature finds that proximity captures the availability of soft information about the firm, which lowers screening and monitoring costs for the bank.²⁴ We also include the duration of the relationship, a standard proxy for relationship intensity; duration is the number of years we observe the bank-firm pair, and it is truncated at a

 $^{^{24}}$ For example Degryse and Ongena (2005) and more recently Agarwal and Hauswald (2010) all find that distance is an important factor in determining credit conditions faced by firms.

maximum value of nine because the reports from which we extract our dataset start in 2005.

We collect variables related to credit risk and other firm characteristics, including profitability, leverage, and liquidity, which banks consider when setting interest rates. We measure profitability as gross operating profits, scaled by total assets (EBITDA Ratio); liquidity as liquid assets scaled by total assets; leverage as the ratio of debt to the sum of debt and equity. Furthermore, to capture credit risk we include a score based on the methodology proposed by Altman (1968), computed by Cerved. The score takes values from one to nine, increasing in credit risk. In particular, we focus on a dummy identifying firms with scores above six, considered risky in the Cerved methodology. To proxy for industry and region-specific characteristics, we include industry dummies based on the two-digit Statistical Classification of Economic Activities adopted by the EU,²⁵ and region dummy variables for the location of the firm's headquarters (North West, North East, Center and South). Finally, we track whether firms own only one credit relationship in good standing or many, to identify captive customers.

Finally, we collect data on banks' characteristics that are likely to influence the cost of loans, particularly funding and capitalization. We construct the following bank variables: the Tier 1 capital ratio, the ratio of liquid assets to total assets, the fraction of assets financed with retail sources, and the fraction of assets financed with wholesale sources, excluding central bank funding. We also include the log of total assets to control for bank size. We use data from the capital section of supervisory reports to compute the difference between banks' transitory CET1 ratios (as of March 2014) and the fully phased-in ones. This measure, whose legal details we explained in Section II.2, will be key in studying bank-level heterogeneity in the impact of the risk-weight discount.

III.3 Data Description

Matching firms from Cerved and loan data from the Credit Register yields approximately 515,000 bank-firms pairs for 2014, of which 236,500 have information on interest rates. Among these, 230,000 are eligible relationships of eligible firms (most Italian firms are SMEs); approximately 6,500 observations are instead non-eligible relationships of eligible

 $^{^{25}}$ See the EuroStat glossary available at https://ec.Europa.eu/Eurostat/statistics-explained/index.php/Glossary:Statistical_classification_of_economic __activities_in_the_European_Community_(NACE) for details.

firms. We also keep data on non-eligible firms to run placebo regressions.

In Figure 2, we show the scatter plot of observations around the two SME-SF assignment thresholds (firms turnover and exposure). The plots show that, although significantly fewer observations refer to large firms with large amounts of disbursed credit (the fourth quadrant in Figure 1), observations' density decreases continuously with size. There are no evident "holes" in our coverage of the treatment space.

We report descriptive statistics regarding SMEs' credit and balance sheet characteristics in Table 1, after the SME-SF implementation (2014, left panel) and before (2013, right panel, a placebo sample). The information on the changes in interest rates refers to average spreads against the quarterly inter-bank rate year-on-year. At the same time, we measure the control variables as of the end of 2013 for the SME-SF implementation sample, and as of the end of 2012 for the placebo sample.

Descriptive statistics for the bank characteristics show that, on average, banks' balance sheets became stronger over time as Liquidity and the CET1 ratio increased, while Retail Funding stayed constant. Although interest rates increased more in 2014 than in 2013, relationship data suggest that only in 2014 the cost of credit increased less on eligible than on non-eligible relationships.²⁶ All firm characteristics stayed similar in the two periods for both firms with and without eligible credit relationships.

Interestingly, the cost of non-eligible lines increased by 11 basis points more than that of the eligible lines in 2014. In 2013 the rate on non-eligible lines increased by 7 basis points less than on eligible lines. We can derive a rough approximation of the impact of the SME-SF, which turns out to be a reduction of around 18 basis points for treated credit lines, quite close to the result we obtain in Section V. Nevertheless, estimating the effect by regression discontinuity is crucial, as the result of this back-of-the-envelope calculation may well reflect differences between large and small lines that are unrelated to the SME-SF.

For example, we can observe that eligible relationships are younger, have a higher share of revolving loans, and have a lower drawn-to-granted ratio. Such heterogeneity may confound our pass-through estimates if not dealt with appropriately. In the next Section, we describe our approach to estimating the causal effects of the SME-SF based

 $^{^{26}}$ The quarterly inter-bank rate sharply decreased at the end of 2014, which explains the especially large difference in the change in spreads. This change does not affect any result, as it gets differenced out in any comparison we perform.

on RDD, and discuss evidence supporting its validity.

IV Empirical Strategy

The ideal experiment to elicit the effect of the SME-SF on the cost of credit would consist of the random assignment of the risk-weight discount to credit relationships of a set of identical firms f borrowing from identical banks b. The difference in the cost of credit between treated and untreated relationships would measure the effect of the SME-SF.²⁷ The design of the SME-SF allows us to approximate this ideal, as we can exploit it for an RDD. The RDD controls for demand confounders affecting different relationships of the same firm and supply confounders affecting different relationships by the same bank. This approach improves on including firm and bank fixed effects and considering all relationships irrespective of their size.²⁸ For example, the flexibility of the RDD approach allows us to directly address the identification concerns grounded in the firm and bank-level heterogeneity highlighted in Paravisini, Rappoport, and Schnabl (2022).

Consider a set of banks b = 1, ..., B who lend to firms f = 1, ..., F; each firm f can borrow from different banks. There are two periods, before and after the introduction of the SME-SF. For each bank-firm relationship, the bank pricing function is:

$$i_{bft} = f\left(M_{bf}, D_{bft}, S_{bft}, R_{bft}\right) \tag{1}$$

where M_{bf} represents all the determinants of the cost of credit that are constant over time and specific to the relationship. D_{bft} collects time-varying credit demand confounders, possibly impacting relationships differently (thus bft); S_{bft} denotes time-varying credit supply confounders. Finally, R_{bft} is the regulatory capital charge, i.e., the amount of regulatory capital the bank b has to set aside at time t on a loan granted to firm f.

Suppose we try to estimate the effect of a change in R_{bft} on the cost of credit by the following linear regression, specified in changes to mute variation due to unobservable, static components:

 $^{^{27}}$ The treatment is the SME-SF; treated observations are credit relationships eligible to the SME-SF policy, and vice-versa for the non-treated.

²⁸ Many papers in empirical banking use firm fixed effects for identification. A seminal work is Khwaja and Mian (2008); other studies are Jiménez et al. (2012), Schnabl (2012), Jiménez et al. (2014), Jiménez et al. (2017).

$$\Delta i_{bf} = \alpha + \beta R_{bf} + \epsilon_{bf} \tag{2}$$

where the effect of the SME-SF would be captured by the coefficient β of a dummy R_{bf} equal to 1 if the risk weight applied to the loan to firm f by bank b at time t = post benefits from the SME-SF (the relationship bf is treated), 0 otherwise (the relationship bf is not treated) and ϵ_{bf} is the residual term. From Equation 1, assuming linear separability in error components, we see that ϵ_{bf} includes three elements: $\epsilon_{bf} = D_{bf}^* + S_{bf}^* + e_{bf}$, where e_{bf} is the true idiosyncratic error component.

The presence of demand and supply factors in the residual may cause bias for different reasons. First, it may be that $\operatorname{cov}\left(S_{bf}^{*}, R_{bf}\right) \neq 0$ as, for example, eligible relationships tend to involve small banks. Since small banks typically do not employ internal risk weighting models (Behn, Haselmann, and Wachtel, 2016), they are more affected by the Basel III reform, and their customers should experience a larger increase in interest rates, perhaps offsetting the benefit from the SME-SF. Consequently, $\hat{\beta}$ would be biased downwards by the non-random matching between firms and banks. Moreover, it is likely that $\operatorname{cov}\left(D_{bf}^{*}, R_{bf}\right) \neq 0$. A firm that borrowed more than 1.5 million before the SME-SF implementation is likely to experience a higher demand for credit than a similar firm that did not. If demand shocks are positively correlated over time, a higher incidence of interest rate increases for firms with non-eligible credit lines will bias upwards β . Finally, even if we focused on firms with multiple relationships, we would include very heterogeneous observations (some very small, some large loans) in the comparison. Banks might be pricing large loans differently, and firms might withdraw more credit from "preferred" relationships in case of demand shocks, while holding some backup credit lines (Detragiache, Garella, and Guiso, 2000; Sette and Gobbi, 2015).

The RDD overcomes the issue of comparability because it is a local approach. It exploits an arbitrary threshold on a continuous variable that defines the treatment status. Suppose that bank-level and firm-level confounding factors do not vary *discontinuously* around the threshold. In that case, we can use the untreated relationships close to the threshold as counterfactual for the treated relationships close to threshold and attribute any discontinuous change in the cost of credit to the SME-SF.

As stated in Section III, eligibility to the SME-SF is based on a bi-dimensional assignment rule (see Figure 1) that takes into account firm's gross sales (turnover) and the credit drawn by the firm from the bank. The turnover threshold is part of the criteria that define an SMEs as per EU law. We thus cannot use it for identification, as other confounding factors vary discontinuously at this threshold. Instead, we focus on SMEs (i.e., firms below such turnover threshold) and implement an RDD around the \in 1.5 million threshold.

Conditional on meeting the turnover eligibility criterion, the treatment probability changes sharply at the threshold:

$$x_{bf} = \text{drawn credit}$$

Eligibility_{bf} = $R_{bf} = \begin{cases} 1 & \text{if } x_{bf}^{2013} \leq \bar{x} = \text{Euro 1.5 million} \\ 0 & \text{otherwise} \end{cases}$

The change from 1 to 0 of the treatment probability defines a sharp RDD. The resulting equation is:

$$\Delta i_{bf} = a + \phi(|x_{bf}^{2013} - \bar{x}|) + \beta R_{bf} + \nu_{bf}$$
Estimated on $bf: x_{bf} \in [\bar{x} - h^-, \bar{x} + h^+]$
(3)

where h^- , h^+ delimit the bandwidth of choice; *a* is a common intercept; $\phi(.)$ summarizes smooth polynomial components in the distance from the threshold; β is the parameter of interest, measuring the effect of the treatment; ν_{bf} is a stochastic error component.

To estimate β in Equation 3, we follow Calonico, Cattaneo, and Titiunik (2014); Calonico et al. (2017) and compute $(\hat{\beta}, \hat{\phi}, h^{+/-})$ minimizing the mean square error of a local polynomial regression. We choose a flexible bandwidth, different on the two sides of the threshold, to account for how observations' density decreases with loan size. Automating the bandwidth choice reduces our degrees of freedom, while Calonico et al. (2017)'s routine ensures that we correct our estimates for the bias introduced by bandwidth selection.²⁹

IV.1 Validity of the RDD Design

The validity of RDD depends on the relatively weak assumptions of continuity of all possible confounders at the assignment threshold and of no perfect manipulation of the

 $^{^{29}}$ In the Appendix, we provide details on the estimation procedure (Section A.1) and show how our choice is conservative, and our results do not depend on it (Section A.2).

assignment variable by treatment takers.³⁰ Given the richness of our data, we can take several steps to show that concerns about the RDD validity are reasonably limited. We provide two types of evidence to support our assumptions. The first is the direct evidence of the lack of assignment variable manipulation (McCrary, 2008); the second is the evidence of continuity of relevant exogenous variables at the threshold (see, e.g. Lee and Lemieux, 2010). Any evidence of manipulation or discontinuity in covariates would raise the concern of sorting around the threshold, invalidating the design.

Manipulation: If the subjects under study knew of the treatment before its introduction and could *perfectly* manipulate drawn credit, they could sort on their preferred side of the threshold. Then, sorting could correlate with unobservables, and these may vary discontinuously at the threshold confounding the policy's effect.

In principle, one could argue that more informed firms anticipated the policy and adjusted their credit demand to stay below the eligibility threshold and benefit from the capital charge discount. If these firms were better managed - they were promptly aware of relevant policy changes - they would also be plausibly able to negotiate lower interest rates for reasons other than the SME-SF. Alternatively, banks that faced a capital shortage might inform their corporate borrowers of the SME-SF, encouraging them to lower their exposure to bring it below 1.5 million.

A first counterargument is that, in practice, the demand for credit of firms is subject to unforeseen shocks that can move marginal credit relationships from one side to the other of the SME-SF eligibility threshold. Exposure (drawn credit) defines eligibility, and the notion of exposure includes contingent liabilities such as guarantees and letters of credit provided by banks. Evidence that firms hold significant amounts of unused credit lines to meet unexpected needs supports the idea that the demand for cash is, to some extent, unpredictable. Indeed, our sample's average ratio of credit disbursed to credit granted is about 60 percent. Perfect manipulation would be difficult and costly. For example, a machine that breaks and needs repairs causes a lumpy need for cash. If the firm must cover the expense with debt, it may not always be able to cut its overall credit demand by the amount that keeps its credit relationship SME-SF compliant. Doing so may imply not executing the repair and delaying production.

 $^{^{30}}$ Manipulation of the assignment variable would imply that manipulators are on one side of the threshold, violating the continuity assumption. For technical details, see Hahn, Todd, and Van der Klaauw (2001).

A second counterargument is that, even if firms could manage their exposure precisely at all times, perfect manipulation would require ex-ante knowledge of the *exact* eligibility threshold. We note that before the approval of the SME-SF, there was considerable uncertainty about the eligibility rules. Although the discussion on the SME-SF began in 2012, regulators initially considered "a reduction by one third of the risk weight for the retail exposure class and an increase of the threshold for retail from \in 1 million to \in 5 million for SMEs" (EBA, 2016). The 1.5 million exposure threshold appeared in the final draft, approved on the 26th of June, 2013,³¹ but banks were uncertain about the criteria they had to follow until 2014.³² We can thus conclude that banks were unlikely to be ready to identify eligible exposures early enough to incentivize many marginally ineligible customers to reduce their exposure below the threshold before the change becoming effective.

We test for manipulation following McCrary (2008) to support our case. When the incentive to manipulate goes in one clear direction, a discontinuity in the density of observations around the threshold should be visible. If firms prefer to be eligible and there are enough informed firms, we should observe significantly fewer marginally noneligible relationships than the marginal eligible ones. A simple density test can highlight a statistically significant drop in the density just above the SME-SF threshold.

We run the test on the drawn credit outstanding density for December 2012, 2013, and 2014. Hence, we search for potential manipulation from the time of the first public mentions of the SME-SF until its implementation *immediate aftermath*. Checking for the aftermath of the policy implementation is important, as borrowers capable of reducing ex post their credit take up to reap the SME-SF benefit would be of arguably greater quality, biasing the treatment effect estimate upwards. The test does not detect any statistically

³¹ The SME-SF The first official record in a "proposal for a regulatimeline is: of the European Parliament and of the Council on prudential requirements for tion credit institutions" dated 12th of June 2012, in which a 2 million limit was discussed (at http://www.Europarl.Europa.eu/sides/getDoc.do?type=REPORT&reference=A7-2012-0171&language=EN#title1); the proposal was assessed by the EBA in September 2012 (EBA, 2012), which focused on the possibility of increasing the retail threshold to $\in 2$ million for banks calculating their capital requirement with the Standard Approach, and to \in 5 million for banks calculating their capital requirement with the Internal Ratings Based Approach; the Commission proposal was then brought to final debate in the European institutions during spring 2013; the reform was finally approved in June 2013.

 $^{^{32}}$ As in Section III, we refer to the EBA Q&A, which included questions submitted until the 27th of November 2013, and to which answers were provided well into the 2nd quarter of 2014 (see the EBA's Q&A at https://eba.Europa.eu/single-rule-book-qa/-/qna/view/publicId/2013_565 and https://eba.Europa.eu/single-rule-book-qa/-/qna/view/publicId/2013_417).

significant discontinuity in the density of observations at the threshold, as shown in the different panels of Figure 3 and by test statistics reported in Table 2.

Such lack of manipulation is not surprising in light of the empirical findings we present in the next Sections. Our estimates suggest that with an average of 26 basis points drop in the cost of credit (see Table 3, Section V), the saving on a credit line manipulated to fall below the \in 1,5 million threshold would stand around \in 3,900 a year.³³ Even a firm at the top of the effects' distribution, say with a previous EBITDA over asset ratio a standard deviation above average and dealing with a bank with a high shadow cost of regulatory capital (see Table 4, Section V, implying a 40 basis points discount for such a relationship), would get a mere \in 6,000 discount on the yearly cost of revolving credit.

We can see that such a number is small by computing the average yearly cost of credit for firms with at least one revolving credit line with drawn credit between the SME-SF threshold and \in 1.6 million as of December 2013. For Italian standards, these are medium-sized firms with multiple credit lines, and their total credit drawn at the firm level is approximately \in 6 million. The average yearly revolving rate for such firms is 8%, and thus the annual average cost of revolving lines hovers around \in 500 thousand. Even in the best-case scenario, a manipulating firm would only save 1.2% of its average yearly cost for revolving lines. Therefore, we are not surprised by the lack of manipulation in the year immediately after the SME-SF implementation.

Discontinuity of covariates: Even in the absence of evidence of manipulation, it could be possible that relationships, firms, or banks with specific characteristics are more likely to appear on one side of the threshold than the other. We estimate a version of Equation 3, replacing the dependent variable with each of the relationship, firm, or bank variables described in Section III, to dispel this doubt. In Figure 4, we plot the discontinuity estimates (black diamonds) and confidence intervals (gray shaded areas) for linear RD specifications targeting different bank, firm, and relationship-level characteristics.³⁴ The results do not support the existence of discontinuities at the SME-SF threshold for any of the variables considered.

³³ This number equals the delta rates we estimate times a \in 1,49 million credit line utilization.

³⁴ We report point estimates in Table A1, including results from a simple comparison of means and a second-degree polynomial, showing that continuity is specification-robust.

V Results

We inspect the behavior of interest rate changes around the SME-SF eligibility threshold. To do so, we show in Figure 5 fit and confidence intervals from local kernel regressions of changes in interest rates on past credit utilization at the relationship-level, in a neighborhood of the SME-SF threshold.³⁵

The plots show that in 2014 interest rates increased on average, most likely because the implementation of Basel III increased the overall cost of credit. More importantly, only in 2014 and for the SMEs sample is there evidence of a discontinuity in the interest rate changes at the policy threshold. In 2014, the cost of credit appears to grow more for relationship not eligible to the SME-SF than for their eligible counterparts. Local kernel regressions on 2013 data, or at placebo thresholds inspected at the same time as the SME-SF implementation, or for non-SME, do not show comparable and statistically significant "jumps" in rates.

This evidence suggests that the policy change had an effect. Still, to get a precise idea of the significance and magnitude of the effect we need to correct our discontinuity estimates and confidence intervals for bandwidth selection bias (Calonico, Cattaneo, and Titiunik, 2014). We display results in the first row of Table 3, using a simple comparison of means (degree 0 polynomial), local linear, and quadratic polynomials.³⁶ Our estimates show a sharp difference in the change in interest rates between eligible and non-eligible relationships for SMEs. The magnitude of the difference is between 20 and 27 basis points and is statistically significant.³⁷

Placebos. Thanks to the rich SME-SF' treatment space, we can check whether we detect any threshold effect where we expect none. The first placebo addresses the possibility that other policies already in place may be affecting relationships below and above the threshold of the SME-SF differently. Several policy interventions have supported access to credit for Italian SMEs. There are two main programs for this purpose, the *Nuovo*

 $^{^{35}}$ We select such a neighborhood employing the mean square error minimization method studied in Calonico, Cattaneo, and Titiunik (2014). We perform the necessary computations in Stata, employing the most recent update of the rdrobust package. We constrain the width of the eligible and non-eligible intervals to be equal for the clarity of the graphical presentation.

³⁶ For arguments in favor of focusing on the results of low degree (first and second) local polynomial specifications, see Andrew and Imbens (2017).

³⁷ In the first subsection of Appendix A.2, Tables A2 and A4, we show respectively that (i) the inclusion of covariates does not affect our result, as expected given the continuity shown in Figure 4 and Table A1, and (ii) the result is preserved and larger in magnitude for very small and hand-picked bandwidths.

Plafond PMI Investimenti and the *Fondo Centrale di Garanzia*. None of such programs, to the best of our knowledge, impinges on the same exposure threshold as the SME-SF.³⁸

As both programs were already active as of December 2013, we check that no other discontinuity at the SME-SF threshold was present for Δi_{bf} in 2012-2013 by repeating the estimation of Equation (3) on the pre-treatment period. In the second line of Table 3, we see that none of the specifications exhibit a statistically significant discontinuity.

The second placebo addresses the concern that there could be some alternative driver of our result concerning small credit relationships. It is unlikely for these relationships to be cheaper or subject to a smaller price increase, as the incidence of fixed costs is greater for smaller loans. Yet we may entertain the possibility that capital-constrained banks see them as less capital-consuming, regardless of the SME-SF. If enough banks treat the \in 1.5 million in terms of past exposure as a rule of thumb for classifying relationships as small, we could have a spurious driver of our results.

If this were the case, though, we should also find a discontinuity at the threshold for firms that are not SMEs according to the specific definition that applies to the SME-SF. Therefore, we run a placebo test estimating Equation (3) on firms with turnover above \in 50 million. We display the results in the third line of Table 3, showing no such discontinuity, no matter the specification.

Bandwidth robustness. We run a further check on the behavior of estimates under different bandwidths. In Figure 6, we show that if we drop the optimal bandwidth selection (and thus bias-correction) and we force small but progressively increasing bandwidths, we converge to the estimate reported in Table 3. Going from a symmetric bandwidth of $\in 25,000$ to one of $\in 205,000$, we see a discontinuity estimate gradually approaching -25 basis points in 2014 and 0 in 2013, with progressively smaller errors. This orderly behavior mitigates the concern that our result is spurious and only due to including odd observations through sample selection.³⁹

V.1 Heterogeneity: Switching Costs and Capital Scarcity

The baseline results capture the average pass-through from the SME-SF capital discount to the cost of credit. This effect might not reflect the full extent of the benefit of a capital

 $^{^{38}}$ We refer to Infelise (2014) for details on such programs.

 $^{^{39}}$ We provide details on the point estimates in Appendix Table A4.

discount for banks. First, the pass-through may be smaller for firms that have difficulties finding alternative sources of credit. Second, the shadow cost of regulatory capital will likely differ across banks, depending on how binding the regulatory constraint is. If our $\hat{\beta}$ captures the shadow cost of regulatory capital, we expect estimates to be greater for banks with a higher shadow cost. As our setting does not require any fixed effects for identification, we are in a unique position to explore such heterogeneity.

In this subsection, we first introduce our proxies of switching costs and regulatory capital scarcity. Then, we demonstrate that both dimensions drive heterogeneity in the SME-SF pass-through to the cost of credit. Moreover, we investigate the effect of the SME-SF on credit growth and find no average local effect from the SME-SF on the growth of credit granted.⁴⁰ Nevertheless, the absence of a quantity effect at the threshold may conceal pass-through heterogeneity. We show that the evidence points in this direction, suggesting that banks reduce granted credit to subsets of eligible borrowers that can be identified as weaker/more bank dependent. These include riskier eligible relationships, relationships belonging to firms either characterized with high past utilization rates or endowed with only one credit line, possibly increasing supply to the rest.

We start from **firm-level heterogeneity** as mapped by switching costs proxies. The importance of bargaining power in bank-firm credit relationships is a classic result (e.g. Rajan, 1992), with abundant empirical evidence in support of its relevance as a driver of credit access and cost (e.g. Detragiache, Garella, and Guiso, 2000; Ioannidou and Ongena, 2010; Santos and Winton, 2019). Moreover, recent evidence shows that the same channel is an important mediator for monetary policy pass-through (Agarwal et al., 2023; Benetton and Fantino, 2021; Scharfstein and Sunderam, 2016). Our results suggest that the same holds for the pass-through of capital regulation, consistently with studies finding that the borrower's capacity to switch to other credit providers limits the exploitation of bank market power in credit relationships (see Ioannidou and Ongena, 2010; Barone, Felici, and Pagnini, 2011; Allen, Clark, and Houde, 2019).

For a firm, the ability to switch does not simply amount to having two or more credit relationships. For example, a firm on the verge of default endowed with many credit relationships may be more captive than one with one relationship but a healthy balance sheet. For this reason, we consider not only the number of bank credit lines but also each

 $^{^{40}}$ We document this in Appendix Table A5.

relationship's past utilization rate, the firm's capacity to generate earnings (EBITDA over assets), and the firm's credit risk rating.

In greater detail, the multiple relationships indicator is a dummy taking value one if a firm has multiple credit relationships in good standing. We track the degree of utilization of credit lines and firms' earning capacity using the standardized lag drawn over granted ratio and the standardized EBITDA over assets ratio. We standardize by subtracting to a variable its population mean and dividing the resulting difference by the original standard deviation. Thus, we can interpret results as the effect of one standard deviation changes around the mean in the variable of interest on the pass-through. Finally, we identify risky firms by a dummy variable equal to one if the risk rating is in the top four notches of the nine-value scale of the rating.

Regarding **bank-level heterogeneity**, we consider a measure of capital scarcity based on the discussion in Section II.2. We construct our variable as the standardized difference between transitory and fully phased-in capital ratios, using the first available information (March 2014):⁴¹

Basel III Gap_{bt} = Transitory $\operatorname{Ratio}_{bt}$ – Full Basel III Phase-In $\operatorname{Ratio}_{bt}$

we assume that the greater the adverse impact of the new definition, the greater the shadow value of an additional euroof regulatory capital to that specific bank.

The Basel III Gap based on supervisory reports is a better measure than the level of the CET1 ratio, as we are interested in capturing the distance from the desired level of equity for each bank. Indeed, each bank's capital ratio is typically higher than the regulatory minimum. As discussed in Repullo and Suarez (2013) and Corbae and D'Erasmo (2021), a bank with a high regulatory capital ratio may be willing to accumulate even more equity and thus have a very high shadow value of capital, and vice versa. As long as the transitory ratio is closer to the bank's desired target than the fully phased-in one, a positive value of the gap indicates that the bank will need to increase regulatory capital (and/or shrink risk-weighted assets) to revert to its desired buffer by the year 2018. Conversely, a negative value indicates that a bank will likely hold too much capital under the new regime.

 $^{^{41}}$ We use the risk-weighted asset as of March 2014 as denominators to obtain the capital ratios employed for the results in Table 4. In the Appendix Tables A6 and A7, we show the robustness of our heterogeneity results to different definitions of the gap variable.

Our implicit assumptions are that the new regime causes a shock, the size of which depends on the initial composition of the bank's capital, and that this shock is independent of each relationship's SME-SF eligibility status. The continuity tests displayed in Figure 4 and Appendix Table A1 (last line) support this last assumption. The difference between the transitory and the fully phased-in CET1 ratios is continuous at the SME-SF assignment threshold.

To obtain a meaningful comparison of parameters and confidence intervals, we estimate the interaction effects in the following local parametric specification:⁴²

$$\Delta y_{bf} = \alpha + \beta_M R_{bf} + \gamma_F \cdot \text{Switching Costs Proxy}_{bft} \cdot R_{bf} + \gamma_B \cdot \text{Basel III Gap}_b \cdot R_{bf} + \Omega X_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + \epsilon_{bf}$$
(4)
Estimated on $bf: x_{bf} \in [\bar{x} - h^-, \bar{x} + h^+]$

where the Switching Costs Proxy is one of the four variables previously introduced; ϕ is a linear polynomial estimated independently on the threshold's two sides; X_{bf} collects controls, which we always include here to mitigate the concern of picking up spurious variations with our interaction coefficients; ϵ_{bf} is an error term clustered simultaneously over firms and banks. Finally, the Δy_{bf} dependent variable stands alternatively for the change in interest rates on the relationship between bank b and firm f and the log change in credit granted by bank b to firm f.

We report in Table 4 the results of estimating Equation 4 for each of the different proxies and with either the price change or the change in the quantity of credit granted as the outcome variable. In the first four columns, we document interest rate change results, and in the last four, those for the difference in the log of credit granted. The first column of each set of regressions reports for reference the estimate of the treatment effect without the inclusion of interaction terms; the second column includes the interaction between the SME-SF and Basel III gap; the third column displays the results obtained by adding the interaction between the SME-SF and the switching costs proxies. Finally, the fourth column jointly shows all interactions. We present four alternative panels, one

 $^{^{42}}$ To perform the estimation, we select the bandwidth using Calonico et al. (2017), construct triangular kernel weights based on such bandwidths, and finally estimate a locally weighted regression employing the Correia (2016) reghdfe package. Of course, the cost of doing so is not correcting point estimates and standard errors for bias as when using Calonico et al. (2017). However, as such correction has a low impact on our main results (see Table A3, which omits the correction), we argue that the scope for concern is limited.

for each switching costs proxy.

We find economically significant pass-through heterogeneity across the first four columns. The interaction coefficients range from 50 to almost 100 percent of the estimates reported in Table 3. Both demand and supply heterogeneity are relevant mediators of the pass-through to interest rates. For example, in the fourth row of the first panel, we note that a standard deviation increase in the drawn over granted ratio almost halves the discount from 28 to 15. A similar result holds for firms with high risk scores, as seen in the last panel's last row.

Conversely, the central two panels show that firms with better outside options drive the pass-through. The last row of the third panel shows that one standard deviation higher EBITDA implies an 11 basis points higher discount. In the second panel, we can see that firms with more than one credit line in good standing drive the SME-SF effect. Finally, in each panel's third row, we show the estimates of the Basel III Gap interaction effect. Across all specifications, banks with one standard deviation larger gap decrease rates to eligible relationships by about six basis points more.

In the second four columns, we study the effect of the SME-SF on credit allocation, measured by the change in the log of credit granted. In the first column of this block, we cannot see any significant average effect on credit granted, i.e., banks do not increase loans to eligible firms more than to other firms. On the contrary, eligible relationships whose utilization rate is higher start being rationed after the introduction of the SME-SF. The effect is economically significant: Credit declines by two percent if the past drawn over granted ratio is one standard deviation larger. A similar result holds for high-risk firms (last panel). In contrast, the second panel shows that the insignificant average effect is probably due to the composition of a decrease in granted credit to firms with only one credit line in good standing with an increase to firms with more than one credit line. Finally, a high Basel III Gap does not influence the pass-through of the SME-SF to quantities.

Overall, our findings on quantities yield support for the policy change introduced in the new EU capital requirement regulation (CRR II), which smoothed the sharp eligibility threshold.⁴³ Indeed, evidence suggests that the sharp discontinuity may have discouraged the extension of credit to creditworthy albeit more fragile customers.

Our results are consistent with the insights from recent theoretical studies showing that firm and bank-level heterogeneity influences the bank lending channel (Ambrocio and Jokivuolle, 2017; Bahaj and Malherbe, 2020; Harris, Opp, and Opp, 2020). In particular, our findings align with Harris, Opp, and Opp (2020), suggesting that adjustments to risk weights disproportionately affect the access to credit of marginal (more creditconstrained) customers.

V.2 What We Learn on the Cost of Capital Regulation to Banks

Our estimates easily convert into a measure of the impact on the cost of credit to the firms from a one percent decrease in the minimum capital requirement. From this impact estimate, under some assumptions, we can learn about the benefit to banks.

Below, how regulators set the minimum capital ratio requirement:

$$\Omega_{bA} = \underbrace{\Theta}_{\text{Minimum Fraction}} * \underbrace{\omega_A}_{\text{Risk Weight}} * A_b$$

here Ω_{bA} is the mandated minimum equity amount bank *b* must set aside given it finances asset *A* for a sum of $\in A_b$.⁴⁴ Ω_{bA} is a Θ fraction of the A_b amount weighted by the ω_A risk weight on assets of type *A*.

Changes in the risk weights cause a change in Ω_{bA} . The eligibility to the SME-SF implies an approximately 2 percentage points saving on the capital required *vis-a-vis* the same exposure without the SME-SF:⁴⁵

⁴³ The new rules foresee an increase in the SME-SF eligibility threshold from € 1.5 million to 2.5 million. Moreover, if the exposure exceeds the threshold, the credit relation will still be eligible for benefits. The fraction below € 2.5 million will still receive the original 76.9 percent support factor. The exceeding portion will enjoy a reduced support factor of 85 percent. See https://eur-lex.Europa.eu/legalcontent/EN/TXT/?uri=CELEX%3A32019R0876.

⁴⁴ In practice, banks hold more than the minimum buffer for prudential reasons, i.e., there exists a $\Theta_b > \Theta$ for each bank b. For a theoretical explanation of such a behavior, see Repullo and Suarez (2013).

 $^{^{45}}$ We use 100 percent as the reference number for the baseline (without SME-SF) risk weight and eight percent as the baseline minimum capital ratio as they are the same employed in the design of the SME-SF itself (see, e.g. EBA, 2016, p.43). They correspond to the one faced by a corporate exposure for a bank that relies on external risk weights (Standard Approach) and does not use an internal risk weighting system for that exposure.

$$\Delta \frac{\Omega_{bA}}{A} = \underbrace{\Theta}_{\text{Minimum Fraction}} * \underbrace{\Delta \omega_A}_{\text{Risk Weight}} \approx -8\% * 24\% = -0.02$$

 \cap

where 24 percent is the approximate decrease in the risk weight on eligible exposures.

The previous sections show that the estimated average impact $\hat{\beta}$ is around 26 basis points. Then, a simple calculation yields the value of the impact on the cost of credit for 1 percentage point change in the minimum capital ratio:

$$\frac{\hat{\beta}}{\Delta \frac{\Omega_{bA}}{A}} = \frac{-26}{-2 \text{ (percentage points)}} =$$

13 bp per percentage point change in the capital requirement

given the heterogeneity results, we stress that this average number is indeed just an average, and the pass-through to better customers should capture the benefit to banks more precisely.

Similarly to Plosser and Santos (2018), we apply the above back-of-the-envelope calculation to a 1 euro loan. The minimum capital requirement on this loan would decrease by 2 cents after the SME-SF implementation. Assuming that the drop in rates for a firm with one standard deviation higher EBITDA thoroughly reflects the initial benefit to the bank from the reform, we sum the first and the last line of column (4) in Table 4's EBITDA panel, and divide the resulting average discount of about 32 basis point by the 2 cent decrease in the requirement per unit of credit. Thus, we derive the shadow cost of 1 additional euro of mandated minimum capital buffer for banks as approximately $16 \notin$ -cents.

Then, interactions with our measure of capital scarcity suggest that the marginal benefit varies with the extent to which each bank is constrained, reaching about $19 \in$ -cents for banks with one standard deviation greater shortfall in regulatory capital resources from an immediate Basel III phase-in (19 is the result of adding the 3 cents greater benefits per euro of requirements' reduction to these banks).

Under the assumption that banks optimally choose their balance-sheet structure, that they are using to the full possible extent every alternative to equity they have, and that they will keep a fixed buffer on top of the minimum requirement – so that 1 euro less in the minimum requirement would imply 1 euro less of equity to hold for the bank – we can read this number as an approximation of the increase in bank profit for holding 1 euro less in equity to finance the loan.

VI Conclusion

We evaluate by an RDD the impact of the 2 percent discount in capital requirements implied by the SME-SF, which applies to SMEs' loans below \in 1.5 million, and find that the cost of eligible loans decreases by approximately 26 basis points. Moreover, we document that the estimated effect is larger for firms more likely to switch to other banks and for banks whose shadow cost of regulatory capital is higher.

Under the assumption of a complete pass-through of the benefit from a lower capital requirement to these low switching costs borrowers, we quantify an approximate 16 bps relief to banks' cost of funding from decreasing the minimum capital buffer by 1 percent, with sizable bank heterogeneity driven by our proxy for regulatory capital scarcity.

Overall, the considerable variation in our estimate of the SME-SF's effect across firms and banks stresses the importance of considering the entire distribution of banks and firms' characteristics to understand who gains or loses from changes in bank capital regulation.

References

- Agarwal, Sumit, Gene Amromin, Souphala Chomsisengphet, Tim Landvoigt, Tomasz Piskorski, Amit Seru, and Vincent Yao. 2023. "Mortgage refinancing, consumer spending, and competition: Evidence from the home affordable refinance program." *The Review of Economic Studies* 90 (2):499–537.
- Agarwal, Sumit, Souphala Chomsisengphet, Neale Mahoney, and Johannes Stroebel. 2017. "Do Banks Pass Through Credit Expansions to Consumers Who Want to Borrow?" The Quarterly Journal of Economics 133 (1):129–190.
- Agarwal, Sumit and Robert Hauswald. 2010. "Distance and private information in lending." *Review of Financial Studies* 23 (7):2757–2788.
- Aiyar, Shekhar, Charles W Calomiris, and Tomasz Wieladek. 2016. "How does credit supply respond to monetary policy and bank minimum capital requirements?" *European Economic Review* 82:142–165.
- Akinci, Ozge and Jane Olmstead-Rumsey. 2018. "How effective are macroprudential policies? An empirical investigation." Journal of Financial Intermediation 33:33–57.
- Allen, Jason, Robert Clark, and Jean-François Houde. 2019. "Search frictions and market power in negotiated-price markets." *Journal of Political Economy* 127 (4):1550–1598.
- Altman, Edward I. 1968. "Financial ratios, discriminant analysis and the prediction of corporate bankruptcy." The Journal of Finance 23 (4):589–609.
- Altunbas, Yener, Mahir Binici, and Leonardo Gambacorta. 2018. "Macroprudential policy and bank risk." Journal of International Money and Finance 81:203–220.
- Ambrocio, Gene and Esa Jokivuolle. 2017. "Should bank capital requirements be less risk-sensitive because of credit constraints?" Research Discussion Paper 10, Bank of Finland. URL https://www.econstor.eu/bitstream/10419/212381/1/ bof-rdp2017-010.pdf.
- Andrew, Gelman and Guido Imbens. 2017. "Why high-order polynomials should not be

used in regression discontinuity designs." Journal of Business & Economic Statistics 35 (2).

- Angrist, Joshua D. and Miikka Rokkanen. 2015. "Wanna get away? Regression discontinuity estimation of exam school effects away from the cutoff." *Journal of American Statistical Association* 110 (512):1331–1344.
- Bahaj, Saleem and Frederic Malherbe. 2020. "The forced safety effect: How higher capital requirements can increase bank lending." *The Journal of Finance* 75 (6):3013–3053.
- Barone, Guglielmo, Roberto Felici, and Marcello Pagnini. 2011. "Switching costs in local credit markets." International Journal of Industrial Organization 29 (6):694–704.
- Basel Committee. 2011. "Basel III: A global regulatory framework for more resilient banks and banking systems." URL https://www.bis.org/publ/bcbs189.pdf.
- Becker, Bo, Marcus M. Opp, and Farzad Saidi. 2021. "Regulatory forbearance in the US insurance industry: The effects of removing capital requirements for an asset class." *Review of Financial Studies*.
- Behn, Markus, Rainer Haselmann, and Paul Wachtel. 2016. "Procyclical capital regulation and lending." The Journal of Finance 71 (2):919–956.
- Benetton, Matteo and Davide Fantino. 2021. "Targeted monetary policy and bank lending behavior." Journal of Financial Economics.
- Berg, Tobias, Markus Reisinger, and Daniel Streitz. 2021. "Spillover effects in empirical corporate finance." *Journal of Financial Economics*.
- Berger, Allen N. and Gregory F. Udell. 1994. "Did risk-based capital allocate bank credit and cause a "credit crunch" in the United States?" Journal of Money, Credit and Banking 26 (3):585–628.
- Berlin, Mitchell and Loretta J. Mester. 1999. "Deposits and relationship lending." Review of Financial Studies 12 (3):579–607.
- Calonico, Sebastian, Matias D. Cattaneo, Max H. Farrell, and Rocio Titiunik. 2017.

"rdrobust: Software for regression discontinuity designs." *Stata Journal* 17 (2):372–404.

- Calonico, Sebastian, Matias D. Cattaneo, and Rocio Titiunik. 2014. "Robust Nonparametric Confidence Intervals for Regression-Discontinuity Designs." *Econometrica* 82 (6):2295–2326.
- Chava, Sudheer and Michael R. Roberts. 2008. "How Does Financing Impact Investment? The Role of Debt Covenants." *The Journal of Finance* 63 (5):2085–2121.
- Claessens, Stijn. 2015. "An overview of macroprudential policy tools." Annual Review of Financial Economics 7:397–422.
- Corbae, Dean and Pablo D'Erasmo. 2021. "Capital buffers in a quantitative model of banking industry dynamics." *Econometrica* 89 (6):2975–3023.
- Correia, Sergio. 2015. "Singletons, cluster-robust standard errors and fixed effects: A bad mix." Technical Note, Duke University.
- ——. 2016. "REGHDFE: Stata module to perform linear or instrumental-variable regression absorbing any number of high-dimensional fixed effects." URL http://scorreia.com/research/hdfe.pdf.
- De Chaisemartin, Clément and Xavier d'Haultfoeuille. 2020. "Two-way fixed effects estimators with heterogeneous treatment effects." *American Economic Review* 110 (9):2964–96.
- Degryse, Hans and Steven Ongena. 2005. "Distance, lending relationships and competition." The Journal of Finance 60 (1):231–266.
- Detragiache, Enrica, Paolo Garella, and Luigi Guiso. 2000. "Multiple versus single banking relationships: Theory and evidence." *The Journal of Finance* 55 (3):1133–1161.
- Diamond, Douglas W. and Raghuram G. Rajan. 2000. "A theory of bank capital." The Journal of Finance 55 (6):2431–2465.
- Drechsler, Itamar, Alexi Savov, and Philipp Schnabl. 2017. "The deposits channel of monetary policy." The Quarterly Journal of Economics 132 (4):1819–1876.

- EBA. 2012. "Assessment of SME proposals for CRD IV/CRR." URL https://eba. europa.eu/documents/10180/16148/EBA-SME-Report.pdf.
- EBA. 2014. "Basel III monitoring exercise." URL https://www.eba. europa.eu/risk-analysis-and-data/quantitative-impact-study/ basel-iii-monitoring-exercise/archived-reports.
- ———. 2016. "EBA Report on Smes and Sme Supporting Factor." URL https://eba.europa.eu/documents/10180/1359456/EBA-Op-2016-04++Report+ on+SMEs+and+SME+supporting+factor.pdf.
- Ganong, Peter and Simon Jäger. 2018. "A permutation test for the regression kink design." Journal of the American Statistical Association 113 (522):494–504.
- Gatzert, Nadine and Hannah Wesker. 2012. "A comparative assessment of Basel II/III and Solvency II." The Geneva Papers on Risk and Insurance-Issues and Practice 37 (3):539– 570.
- Glancy, David and Robert Kurtzman. 2022. "How do capital requirements affect loan rates? Evidence from high volatility commercial real estate." The Review of Corporate Finance Studies 11 (1):88–127.
- Hahn, Jinyong, Petra Todd, and Wilbert Van der Klaauw. 2001. "Identification and estimation of treatment effects with a regression-discontinuity design." *Econometrica* 69 (1):201–209.
- Harris, Milton, Christian Opp, and Marcus M. Opp. 2020. "The aggregate demand for bank capital." Working Paper 27858, National Bureau of Economic Research. URL https://www.nber.org/papers/w27858.
- Indarte, Sasha. 2023. "Moral Hazard versus Liquidity in Household Bankruptcy."
 Forthcoming at The Journal of Finance. URL https://sashaindarte.github.io/
 research/indarte_bk.pdf.
- Infelise, Federico. 2014. "Supporting Access to Finance by SMEs: Mapping the initiatives in five EU countries." Research Report 9, Brussels: European Capital Markets Institute. URL http://europa.eu/rapid/press-release_MEMO-13-272_en.htm.

- Ioannidou, Vasso and Steven Ongena. 2010. ""Time for a change": Loan conditions and bank behavior when firms switch banks." *The Journal of Finance* 65 (5):1847–1877.
- Jensen, Michael C. and William H. Meckling. 1976. "Theory of the firm: Managerial behavior, agency costs and ownership structure." Journal of Financial Economics 3 (4):305–360.
- Jiménez, Gabriel, Steven Ongena, José-Luis Peydró, and Jesús Saurina. 2012. "Credit Supply and Monetary Policy: Identifying the Bank Balance-Sheet Channel with Loan Applications." American Economic Review 102 (5):2301–2326.
- ———. 2014. "Hazardous times for monetary policy: What do twenty-three million bank loans say about the effects of monetary policy on credit risk-taking?" *Econometrica* 82 (2):463–505.
- Jiménez, Gabriel, Steven Ongena, José-Luis Peydró, and Jesus Saurina Salas. 2017. "Macroprudential policy, countercyclical bank capital buffers and credit supply: Evidence from the Spanish dynamic provisioning experiments." Journal of Political Economy 125 (6):2126–2177.
- Kareken, John H. and Neil Wallace. 1978. "Deposit insurance and bank regulation: A partial-equilibrium exposition." *Journal of Business* :413–438.
- Kashyap, Anil K., Jeremy C. Stein, and Samuel Hanson. 2010. "An analysis of the impact of 'substantially heightened' capital requirements on large financial institutions." Booth School of Business, University of Chicago, mimeo 2.
- Keeley, Michael C. 1990. "Deposit insurance, risk and market power in banking." American Economic Review :1183–1200.
- Keys, Benjamin J., Tanmoy Mukherjee, Amit Seru, and Vikrant Vig. 2010. "Did securitization lead to lax screening? Evidence from subprime loans." *The Quarterly Journal* of Economics 125 (1):307–362.
- Khwaja, Asim Ijaz and Atif Mian. 2008. "Tracing the Impact of Bank Liquidity Shocks: Evidence from an Emerging Market." *American Economic Review* 98 (4):1413–1442.

- Kisin, Roni and Asaf Manela. 2016. "The shadow cost of bank capital requirements." *Review of Financial Studies* 29 (7):1780–1820.
- Lecarpentier, Sandrine, Mathias Lé, Henri Fraisse, and Michel Dietsch. 2020. "Lower bank capital requirements as a policy tool to support credit to SMEs: Evidence from a policy experiment." Working Paper 789, Banque de France. URL https://ssrn.com/ abstract=3771271.
- Lee, David S. and Thomas Lemieux. 2010. "Regression discontinuity designs in economics." *Journal of Economic Literature* 48 (2):281–355.
- Mayordomo, Sergio and María Rodríguez-Moreno. 2018. "Did the bank capital relief induced by the Supporting Factor enhance SME lending?" *Journal of Financial Intermediation* 36:45–57.
- McCrary, Justin. 2008. "Manipulation of the running variable in the regression discontinuity design: A density test." *Journal of Econometrics* 142 (2):698–714.
- Miles, David, Jing Yang, and Gilberto Marcheggiano. 2012. "Optimal bank capital." The Economic Journal 123 (567):1–37.
- Myers, Stewart C. and Nicholas S. Majluf. 1984. "Corporate financing and investment decisions when firms have information that investors do not have." *Journal of Financial Economics* 13 (2):187–221.
- Paravisini, Daniel, Veronica Rappoport, and Philipp Schnabl. 2022. "Specialization in Bank Lending: Evidence from Exporting Firms." Journal of Finance (forthcoming) URL https://pages.stern.nyu.edu/~pschnabl/research/PRS_2022.pdf.
- Peek, Joe and Eric S. Rosengren. 2000. "Collateral Damage: Effects of the Japanese Bank Crisis on Real Activity in the United States." *American Economic Review* 90 (1):30–45.
- Petersen, Mitchell A. and Raghuram G. Rajan. 1995. "The effect of credit market competition on lending relationships." *The Quarterly Journal of Economics* 110 (2):407–443.
- Plosser, Matthew C. and João A. C. Santos. 2018. "The cost of bank regulatory capital." Staff Reports 853, Federal Reserve Bank of New York. URL https://papers.ssrn. com/sol3/papers.cfm?abstract_id=3199903.

- Rajan, Raghuram G. 1992. "Insiders and outsiders: The choice between informed and arm's-length debt." The Journal of Finance 47 (4):1367–1400.
- Repullo, Rafael and Javier Suarez. 2013. "The procyclical effects of bank capital regulation." Review of Financial Studies 26 (2):452–490.
- Rodano, Giacomo, Nicolas Serrano-Velarde, and Emanuele Tarantino. 2018. "Lending standards over the credit cycle." *Review of Financial Studies* 31 (8):2943–2982.
- Santos, João A. C. and Andrew Winton. 2008. "Bank loans, bonds and information monopolies across the business cycle." The Journal of Finance 63 (3):1315–1359.
- ——. 2019. "Bank capital, borrower power, and loan rates." *Review of Financial Studies* 32 (11):4501–4541.
- Scharfstein, David and Adi Sunderam. 2016. "Market power in mortgage lending and the transmission of monetary policy." Manuscript, harvard business school, Harvard University. URL http://www.people.hbs.edu/asunderam/Mortgage%20Market%20Power% 2020160829.pdf.
- Schnabl, Philipp. 2012. "The international transmission of bank liquidity shocks: Evidence from an emerging market." The Journal of Finance 67 (3):897–932.
- Sette, Enrico and Giorgio Gobbi. 2015. "Relationship lending during a financial crisis." Journal of the European Economic Association 13 (3):453–481.
- Sharpe, Steven A. 1990. "Asymmetric information, bank lending and implicit contracts: A stylized model of customer relationships." *The Journal of Finance* 45 (4):1069–1087.
- Thistlethwaite, Donald L. and Donald T. Campbell. 1960. "Regression-discontinuity analysis: An alternative to the ex post facto experiment." *Journal of Educational Psychology* 51 (6):309–317.
- Wang, Yifei, Toni M. Whited, Yufeng Wu, and Kairong Xiao. 2020. "Bank market power and monetary policy transmission: Evidence from a structural estimation." Working Paper 27258, National Bureau of Economic Research. URL https://www.nber.org/ papers/w27258.pdf.

		2014			2013							
	Mean	Std Dev.	Count	Mean	Std Dev.	Count						
	Ν	Ion-Eligib	le Credit	Relation	ships							
Δi_{bf}	37.02	200.76	6,510	-3.293	221.18	8,901						
Drawn	3,069.94	$3,\!697.87$	6,510	3,032.69	$3,\!674.05$	8,901						
Granted	$3,\!931.78$	$4,\!479.97$	6,510	$3,\!510.72$	4,204.72	8,901						
Drawn/Granted	81.151	17.695	6,509	82.428	17.489	$8,\!615$						
Revolving F.	17.9	28.8	6,509	17.2	28.3	$8,\!615$						
Age Rel.	7.202	2.534	6,510	6.369	2.352	8,901						
	Firms	With Only	y Non-E	ligible Rel	ationships							
Sales	$3,\!809.64$	$7,\!917.63$	6,985	$3,\!860.6$	7,925.29	$8,\!656$						
Leverage	73.206	31.670	$6,\!983$	72.797	32.01	$8,\!653$						
EBITDA	3.413	6.575	6,985	3.489	6.356	$8,\!656$						
Risk Score	5.498	1.673	6,587	5.467	1.686	8,190						
N. Relations	1.900	1.383	6,985	1.895	1.351	$8,\!656$						
	Eligible Credit Relationships											
Δi_{bf}	26.22	180.12	229,871	4.61	186.6	252,586						
Drawn	216.65	253.12	229,871	226.9	261.2	252,586						
Granted	390.66	473.64	229,871	386.7	506.96	252,586						
Drawn/Granted	58.795	32.045	229,667	59.709	32.101	252,586						
Revolving F.	32.3	33.3	229,667	0.322	0.333	252,586						
Age Rel.	5.432	3.123	$229,\!871$	4.988	2.759	$252,\!586$						
	Firms V	Vith At Le	east One	Eligible H	Relationship							
Sales	2,710.57	$5,\!143.67$	170,283	$2,\!688.78$	5,089.24	185,697						
Leverage	57.721	33.937	$170,\!230$	58.676	34.126	$185,\!639$						
EBITDA	6.795	8.993	170,278	6.547	9.085	$185,\!697$						
Risk Score	5.192	1.734	162,704	5.265	1.732	$177,\!944$						
N. Relations	2.666	1.723	$170,\!283$	2.680	1.727	$185,\!697$						
			Banks	5								
CET1 Ratio	12	3.6	90	11.7	3.5	94						
Retail Funding Ratio	62.2	15.6	90	62.2	16.3	94						
Liquidity Ratio	20.8	8.6	90	16.1	7.4	94						
Basel III CET1 Gap	0.1	1.4	61									

Table 1: **Descriptives**

Note: A "relationship" is a bank-firm pair, reporting the total exposure firm f has toward bank b. The loan-level data comprise all performing loans from Italian banks in good standing (for which we have complete balance sheet information), to Italian firms with available CERVED balance sheet data. The first three columns of the Table report information on mean, dispersion and count for the year 2014, and we measure all variables as of the end of year 2013, except for the change in the interest rate, which averages 2014 quarterly changes. The second three columns of the table report information on mean, dispersion and count for the year 2013, and we measure all variables as of the end of year 2013 quarterly changes. Δi_{bf} measures the change in the interest rate, which averages 2013 quarterly changes. Δi_{bf} measures the changes in yearly revolving rates in basis points; we report all ratios in percentage points. The CET1 gap variable is the difference between the transitory and full-Basel III-phase-in CET1 ratios. We only report information for the sub-sample of observations for which we can observe interest rates.

	SME End of 2014	SME 2014	Non-SME 2014	SME 2013
Observations (l - r) Optimal Bandwidths	822 — 492 98.25 - 58.57	746 - 463 90.921 - 58.42	42 - 74 34.06 - 34.34	689 - 595 69.24 - 56.26
t-Statistics p-Values	-0.051 0.959	$0.708 \\ 0.479$	$1.49 \\ 0.136$	$0.82 \\ 0.411$

Table 2: McCrary's Density Test for outstanding exposure

Note: The Table presents the t-statistics and p-values of the McCrary's density test. We report in the first row the number of observations we consider for density estimation at the left and right of the cutoff. In all cases, the null hypothesis is that there is no discontinuity in the density. We optimally select bandwidths minimizing the MSE of the density estimates, and do so independently on the two sides of the threshold. We report bandwidth limits in thousand of Euros.

	RD, $Pol(0)$	RD, Pol(1)	RD, $Pol(2)$
\hat{eta} 2014	-19.785^{***} (6.458)	-25.949^{***} (8.202)	-26.991^{***} (10.138)
Obs. (left; right)	4,816; 3,181	9,450; 5,675	17,948; 6,232
MSE-Optimal Bdw.	447.62 - 695.61	659.72 - 2,811.44	981.32 - 5,649.42
\hat{eta} 2013	0.812 (5.2)	1.595 (6.602)	$1.599 \\ (7.959)$
Obs. (left; right)	10,326; 5,021	$24,511;\ 7,469$	37,293; 8,204
MSE-Optimal Bdw.	618.58 - 967.23	930.34 - 2,883.47	1,063.78 - 5,551.15
$\hat{\beta}$ 2014 (Non-SME)	-6.87 (16.642)	-6.562 (20.285)	1.204 (29.56)
Obs. (left; right)	328; 2,774	701; 3,825	686; 4,126
MSE-Optimal Bdw.	237.15 - 3,657.1	496.66 - 10,802.43	487.75 - 20,095.03

Table 3: Dependent Variable Interest Rate Change in bp; Method Simple RD

Note: This Table presents the results of discontinuity tests run *via* local estimation of $\hat{\beta}$ in Equation 3: $\Delta i_{bf} = \alpha + \beta R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + \nu_{bf}$, where Δi is the interest rate change in basis points, x_{bf} the past drawn credit, \bar{x} the \in 1.5 million threshold, ϕ the x_{bf} polynomial independently estimated on the two sides of the threshold, and the null hypothesis of each test is $\beta = 0$. The different columns report increasing polynomial specifications. We obtain estimates for the SMEs 2014 sample, for the SMEs 2013 and non-SMEs 2014 placebo samples. Estimates reported employ triangular kernel weights, with robust standard errors displayed in parentheses. We report observations left and right of the cutoff and the corresponding optimal bandwidths (in thousands of Euros) below each estimates' block.

	Dep. Variable:		Rates Cha	ange, bps			Granted,	Log Chan	ge
		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	SME-SF	-20.91*** (-3.10)	-18.92** (-2.66)	-29.06*** (-3.87)	-27.81*** (-3.46)	$.0015 \\ (0.15)$.0022 (0.21)	.0145 (1.05)	.0168 (1.12)
Drawn/	Basel III Gap		-20.6*** (-5.40)		-20.62*** (-5.43)		0088*** (-3.15)		0087*** (-3.09)
Granted Panel	Basel III Gap * SF		-5.601*** (-3.24)		-5.646*** (-3.24)		0014 (-0.44)		0016 (-0.53)
	Drawn/Granted			8.154 (1.17)	8.946 (1.23)			.0215** (2.28)	.0231** (2.35)
	Drawn/Granted * SF			(2.14)	12.76^{**} (2.41)			0179** (-2.09)	02** (-2.22)
	SME-SF			8016 (-0.06)	.9183 (0.07)			0391** (-2.24)	0472^{**} (-2.57)
Relationships	Basel III Gap				-20.68*** (-5.40)				0087^{***} (-3.11)
Panel	Basel III Gap * SF				-5.434*** (-3.19)				0018 (-0.59)
	Multi Rel.			3.807 (0.39)	2.824 (0.24)			0089 (-0.91)	01 (-0.97)
	Multi Rel. * SF			-21.16** (-2.09)	-20.84* (-1.75)			.0443*** (3.29)	$.0537^{***}$ (3.90)
	SME-SF			-23.41*** (-3.53)	-21.69*** (-3.05)			.0024 (0.24)	.0036 (0.34)
EBITDA	Basel III Gap			· /	-20.39*** (-5.38)			. ,	0089*** (-3.19)
Panel	Basel III Gap * SF				-5.905*** (-3.37)				-0011 (-0.37)
	EBITDA			-3.55 (-1.11)	-3.488 (-1.08)			$.0221^{***}$ (3.42)	$.0216^{***}$ (3.26)
	EBITDA * SF			-9.819** (-2.24)	-10.98** (-2.49)			.0049 (0.88)	.0078 (1.41)
	SME-SF			-25.97***	* -24.73***			.0162	.0185
Risk	Basel III Gap			(-3.88)	(-3.49) -20.54*** (-5.40)			(1.46)	(1.55) 009*** (-3.27)
Panel	Basel III Gap * SF				-5.886*** (-3.30)				-1.8e-04 (-0.06)
	Log Risk Score			17.63 (1.58)				.0041 (0.34)	.0047 (0.37)
	High Risk * SF			12.51^{***} (4.72)	14.51^{***} (5.46)			0426*** (-4.83)	0477*** (-5.10)
	Linear Rel., Firm, Bank Contro	√ ols √	√ √	\checkmark	√ √	√ √	√ √	\checkmark	√ √
	Observations	14.64	4 13.817	14,644	13,817	18,21	2 17.059	18,212	17.059

Table 4: The Role of Bank Capital and Credit Scarcity

Note: This Table explores how credit scarcity, proxied by different relationship or firm-level characteristics, and bank capital scarcity, proxied by the difference between bank's CET1 under the transitional and full Basel III phase-in definitions (Basel III Gap), interact with the SME-SF pass-through. The first four columns report effects on rates; first, the baseline effect of a local linear specification, for reference; second, the interaction between SME-SF Eligibility and bank capital scarcity; third, the interaction with credit scarcity; fourth, the model including both interaction terms. The second four columns report effects on granted credit. The four different panels use alternative proxies for credit scarcity. We estimate all specifications locally with triangular kernel weights, over bandwidths chosen to minimize MSE. t-Statistics, from errors clustered at the firm and bank level, are in parentheses. *Controls*: **Relationship-level**: Lags of share of total drawn credit, revolving granted/total granted, utilized/granted, firm's and bank's hq in same province, log(relationship age). **Firm-level**: Lags of liquidity ratio, leverage, log(assets), log risk score (Altman z-score), EBITDA/assets, industry dummies, regional dummies, dummy for the presence of multiple relationships, investment ratio. **Bank-level**: A dummy singling out cooperative banks; the lag of CET1, liquidity, retail and wholesale funding ratios; the log of lag assets. We report robustness to alternative proxies of capital scarcity in Tables A6 and A7.





EUR 50 mn

Note: The Figure presents the assignment space defined by the SME-SF eligibility rules.

Figure 2: Observations in the Treatment Space



Note: The Figures present the distribution of bank-firm relationships over the treatment space.



Figure 3: Tests for the Discontinuity in Observation Density

Note: The Figures present the graphical outputs of testing for discontinuity in the density of observations on the left and the right of the cutoff. The error bars plot the 95 percent confidence interval around the density point estimates. The first panel (top-left) reports the test for SMEs, using credit drawn at the end of 2013 as a running variable; the second (top-right) reports the test for SMEs, using credit drawn at the end of 2012 as a running variable; the third (bottom-left) reports the test for SMEs, using credit drawn at the end of 2014 as a running variable; the fourth (bottom-right) reports the test for Non-SMEs, using credit drawn at the end of 2013 as a running variable; the fourth (bottom-right) reports the test for Non-SMEs, using credit drawn at the end of 2013 as a running variable.

Continuity Tests RD Coefficients						• Po	int e	stim	ate		95%	6 CI		90	% CI					
.3																				
0 -		•		•	•	•	-	•			•	٠		•		•	•	•	•	• —
3															,					
Variable																				
Lag Share Total Drawn																				•
Lag Revolving Share																				
Lag Drawn on Granted																		•		
Lag Revolving Rate																	•			
Age																•				
HQ in Same Province													•		•					
Lag Leverage		•											•	•						
Lag EBITDA/Assets		•											•							•
Lag log(Assets)		•		•								•	•							
Multiple Rel.		•				٠					•	•								
Lag Liquidity		•		•						•										
Lag Investment									•											
Risk Score								•												
Lag Tier 1 Ratio		•					•													
Lag Liquidity		٠	٠			•							٠							
Lag Retail Funds		٠	٠		•								•							
Lag Whole Funds		٠	٠	•									•							
Lag Bank Size			•		•															
BCC		٠																		
Basel III Gap	•	٠	٠		٠								•	•	•	•				•

Figure 4: SME-SF Discount Assignment

Note: The Figure reports the point estimates (black diamonds) and confidence interval (gray shaded areas) for discontinuities in relationship, firm and bank-level characteristics. We estimate the following specification covariate_{bf2013} = $b_0 + b_1 R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + e_{bf}$ within optimally chosen bandwidths, with a triangular kernel. Here x_{bf} is drawn credit, \bar{x} the \in 1.5 million threshold, ϕ the linear x_{bf} polynomial independently estimated on the two sides of the threshold, and b_1 the parameter of interest we estimate and plot. Below the graph, under the label "Variable", each line marks which variable we test for discontinuity. A black dot to the right of the corresponding line marks that the above estimate and confidence interval concern that specific covariate.



Figure 5: Discontinuity Plots, Reform and Placebos

Note: From the top left, we report the reform effect at the eligibility threshold for SME credit lines in 2013-2014; the placebo for SME credit lines at the SME-SF threshold in 2012-2013; the placebo discontinuity employing non-SME credit lines in 2013-2014; the placebo for the fictitious \in 500 thousand of past utilization threshold, for SMEs in 2013-2014. The figure plots on the y-axis the delta in yearly rates before and after SME-SF implementation (and for the 2012-2013 window in subfigure (b)); on the x-axis, we plot the lag of credit drawn, in thousands of \in . The overall limits of the x-axis shown are selected minimizing the MSE of the discontinuity point estimate, under the constraint of equal spans on the two sides of the threshold for presentation clarity. Gray balls represent binned averages of the data, with ball dimension increasing in the number of observations in each equally spaced bin. Dark (right-of-threshold) and ocra (left-of-threshold) solid lines are smooted polynomial estimates of the relationship between past drawn amounts and rate changes, while dotted lines are the 95% confidence intervals.



Figure 6: Bandwidth Robustness

Note: The Figure reports the point estimates (black diamonds) and confidence intervals (gray shaded areas) for discontinuities at the SME-SF assignment threshold obtained under increasingly larger bandwidths. We estimate the following specification $\Delta i_{bf} = a + \beta R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + \nu_{bf}$, where x_{bf} is drawn credit, \bar{x} the \in 1.5 million threshold, ϕ the linear x_{bf} polynomial independently estimated on the two sides of the threshold, and β the parameter of interest we estimate and plot. Below the graph, under the label "Bandwidth", each line marks the furthest observation to the right and left of the threshold we include in the local estimation. Below the "Year", each line marks which sample we employ, whether concerning the year of treatment (2014), or the year before. A black dot to the right of the corresponding line marks that we computed the estimate above using that specific bandwidth and sample. We report details on the point estimates up to bandwidth \in 105 thousand in Table A4.

A Appendix

A.1 Estimation Details

To estimate β in Equation 3, we model the expected value of interest rate changes separately below and above the SME-SF threshold, as:

$$E[\Delta i_{bf}(1)|x_{bf}] = a^{-} + \phi^{-}(|x_{bf}^{2013} - \bar{x}|) \quad 1,500 - h^{-} \le x_{bf}^{2013} < 1,500$$

$$E[\Delta i_{bf}(0)|x_{bf}] = a^{+} + \phi^{+}(|x_{bf}^{2013} - \bar{x}|) \quad 1,500 \le x_{bf}^{2013} \le 1,500 + h^{+}$$
(5)

Then, using the Stata routine based on Calonico, Cattaneo, and Titiunik (2014) and described in Calonico et al. (2017), we choose $(\hat{a}^{+/-}, \hat{\phi}^{+/-}, h^{+/-})$ minimizing

$$\sum_{bf} \left(\Delta i_{bf} - a^{-} - \phi^{-} (|x_{bf}^{2013} - \bar{x}|) \right)^{2} K \left(\frac{|x_{bf}^{2013} - \bar{x}|}{h^{-}} \right) \text{ below threshold}$$

$$\sum_{bf} \left(\Delta i_{bf} - a^{+} - \phi^{+} (|x_{bf}^{2013} - \bar{x}|) \right)^{2} K \left(\frac{|x_{bf}^{2013} - \bar{x}|}{h^{+}} \right) \text{ above threshold}$$
(6)

where K is a triangular kernel weight function. Consequently:

$$\hat{\beta} = \hat{a}^- - \hat{a}^+ \tag{7}$$

A.2 Robustness of the Results

Covariates' continuity is robust to degree 0 and 2 polynomials specifications: In Table A1, we show that the absence of statistically detectable discontinuities in other covariates, shown in Figure 4, does not depend on the polynomial degree of choice. Moreover, we report in this Table the exact estimates for reference.

Controls inclusion: We repeat the estimation, including bank, firm, and relationship characteristics as further independent variables for robustness. Even though these controls do not vary discontinuously at the eligibility threshold (as we have shown in Section IV.1), their inclusion can increase precision and also provide information on the effect of heterogeneity in observable characteristics on our *average* pass-through effect. As suggested by Angrist and Rokkanen (2015), including control variables mitigates concerns of lack of external validity of RDD estimates and further limits the bias due to the inclusion of observations far from the threshold when the bandwidth is wide. Including covariates brings far-off observations to a more equal footing.

Furthermore, our relationships are stratified at the firm and bank levels. The main correlation concern is at the firm level, as it is reasonable to think that the same team makes decisions on the pricing of loans of the same firm based on the same set of information (e.g., leverage, profitability, credit score). For this reason, we cluster the standard errors at the firm level. We thus obtain the results displayed in Table A2. The estimated effects are similar to our baseline results without including covariates - a discontinuity of 23 to 25 basis points - while statistical significance remains unchanged.

The role of bias adjustment: Then, as the interaction results in display play an essential role in our argument, and as we employ the same bandwidths but cannot correct for the bandwidth selection bias in that case, we must make sure that such adjustment does not play an important role in our main result magnitudes. For this purpose, we report results omitting bias correction in Table A3, where we can see that the results change but for a few basis points.

Hand-picked bandwidths: We show the broad robustness of our result to simple mean comparisons in extremely restricted neighborhoods of the threshold. To do so, we drop the optimal bandwidth selection algorithm (and thus bias-correction) and force alternative bandwidths moving from a symmetric distance of $\leq 25,000$ from the threshold to a $\leq 105,000$ distance at increments of $\leq 10,000$. We can see in Table A4 that the results are larger in magnitude, still significant albeit expectedly noisier, and consistent with a discount caused by the SME-SF, while the 2013 placebo is still small and insignificant.⁴⁶

Relationship between the estimate and the threshold: In Figure A1, we check how the SME-SF's estimate changes if Relationship between the estimate and the threshold: In Figure A1, we check how the SME-SF's estimate changes if we use treatment assignment cutoffs progressively further from the true one. In RDD settings, spurious results may arise from a non-linear relationship between the assignment and the dependent variables. The sharp discontinuity estimator may pick such continuous nonlinear relationships up and "read" them as discontinuities. If such a problem drives our result, we would expect the significance and magnitude of the estimate not to peak at the assignment threshold. We check for this behavior in the same spirit as the permutation test proposed in Ganong and Jäger (2018) for regression kink designs. In the Figure,

⁴⁶ We lack sufficient density to estimate the Non-SMEs placebo on such restricted bandwidths reliably.

we see that the estimate's magnitude and significance peaks at \in 1.5 million, turning non-significant and smaller as we progressively use assignment thresholds further from the true one.

Lack of an average SME-SF impact on granted credit: In Table A5 we show that estimating Equation 6 using changes in lag granted credit as the left-hand side variable delivers a null result.

Robustness of the heterogeneity result to different Basel Gap measures: In Tables A6 and A7, we show that our heterogeneity study is robust to measuring heterogeneity in the banks' shadow cost of capital with alternative proxies of the regulatory capital shortfall induced by Basel III. First, Table A6 displays results employing Tier 1 capital instead of CET1 capital. Tier 1 capital is a broader class encompassing CET1 and other equity-like instruments. A bank that experiences a large Tier 1 wipeout should also have a high shadow cost of regulatory capital. Indeed, results are not affected.

In Table A7, we instead explore what happens with a different measure of CET1 shortfall. The measure we employ in the main body of the paper looks at the difference between transitory and full phase-in CET1 ratios in March 2014, the first date banks started providing such information. The denominator of both ratios is the risk-weighted assets as of March 2014. Employing the same denominator for both ratios, we effectively mute the impact of changes in risk-weighting introduced by Basel III (including the SME-SF). Here, we construct a measure that encompasses the effect of changing risk weights on the target ratios. Instead of using the transitory CET1 ratio as of March 2014, we employ the reconstructed CET1 ratio in December 2013, whose denominator is the 2013 risk weights under the pre-Basel III rules. Then, we subtract the fully phased-in CET1 ratio using March 2014 risk-weighted assets as the denominator. In the Table, we can see that this has no material effect on our results.

A.3 On The Shortcomings of Within Identification

We show the large extent of heterogeneity in the effect of the SME-SF, likely led by switching costs and bargaining power in credit relationships. Moreover, we see that the availability of a backup source for the firm's credit demand - i.e., multiple relationships and a low utilization ratio - is one key factor determining the magnitude of the passthrough. This Appendix further explores this aspect, simultaneously showing a possible pitfall of using the classic Khwaja and Mian (2008) fixed-effect strategy in this and similar contexts.

Using firm fixed effects implies identifying the treatment effect using the within estimator considering the sample of firms with multiple credit relationships. Unfortunately, this procedure can significantly alter the estimated coefficient, even without demand bias, through sample selection. In our case, a firm-fixed effects identification strategy would select a sub-sample of firms with outside options, i.e., firms that can better capture a larger share of the surplus generated by the SME-SF.

Implementing a within RD estimation with high dimensional fixed effects requires some adjustment to the estimation procedure. To perform the within RD, we select the bandwidth using Calonico et al. (2017), construct triangular weights based on such bandwidths, and finally estimate a weighted fixed effect regression using the routine described in Correia (2016), useful to handle high dimensional fixed models.⁴⁷ In Table A8, we can observe how the magnitude of the point estimates increases to values ranging between 35 and 39 basis points, still highly significant.

As we have extensively shown no discontinuous change in a host of observables at the discontinuity threshold, the most likely reason for the difference in the point estimates is the sample selection imposed by the fixed effect estimator. To be sure that this is why we observe increased point estimates, we estimate the model using the same subset of observations exploited by the fixed effect estimator but omitting the fixed effects. We run local regressions using observations belonging to firms with at least two relationships, one eligible for the SME-SF and one not, in the neighborhood of the eligibility threshold selected through the data-driven algorithm. We present the results of such estimation in the first panel of Table A9. The estimated effect is larger than our main result and similar to the one obtained with the fixed effects model, confirming our hypothesis.

Two possible reasons may cause the larger magnitude of the estimated effect for the fixed effects sub-sample. On the one hand, it may be the case that the rates on eligible credit lines of such firms grow less; on the other hand, it may also be true that the rates on non-eligible relationships of such firms rise more. We thus check that the increase in

⁴⁷ We make this choice as the rdrobust Stata routine (Calonico et al., 2017) does not provide a way to handle high dimensional fixed effects directly. To keep working within the rdrobust framework, one should create thousands of firm identifier dummies and feed them to the model, manually dropping local singleton observations for clustered error cases (Correia, 2015). As the reghdfe performs all such steps automatically, it is the least ad-hoc option at our disposal.

the estimated SME-SF impact is not due to a higher increase in the rates of non-eligible credit relationships in the fixed effects sub-sample. In the second panel of Table A9, we show the result of comparison in rate changes for non-eligible credit relationships of firms in and out of the fixed effects sub-sample. Across different specifications, we can see that firms in the sub-sample experience rate changes in line with other firms. We can thus conclude that the SME-SF effect on eligible relationships appears to be stronger in the firm fixed effects sub-sample *independently* of the inclusion of fixed effects.

The fact that sample selection from the fixed-effects strategy is enough to see an increase in the result substantiates our interpretation of the increase in the magnitude of point estimates as coming from the higher bargaining power of firms in the fixed-effects subsample. If firms borrowing from a single bank were locked in a monopolistic relationship with their lender, the latter would not necessarily transfer the benefit from the SME-SF to the firm. Instead, the pass-through would be more significant for firms that can switch between existing relationships, limiting banks' capacity to extract rents.⁴⁸

The sub-sample on which we identify the local fixed effect estimator of the treatment effect is composed of firms with multiple *similar* relationships, at least one of which is eligible and one not. Hence, they are precisely the firms that are less likely to be captured by a relationship lender, as they have other credit relationships that are close substitutes.

 $^{^{48}}$ Again, as argued in Rajan (1992), Detragiache, Garella, and Guiso (2000) and Ioannidou and Ongena (2010).

Control Variable	Test, $Pol(0)$	Test, $Pol(1)$	Test, $Pol(2)$
Lag Share of Total Drawn	-0.002	-0.004	-0.004
	(0.012)	(0.011)	(0.015)
Lag Revolving Fraction	0.014	0.016	0.024
	(0.014)	(0.016)	(0.016)
Lag Drawn on Granted	-0.002	-0.003	0.001
	(0.01)	(0.012)	(0.013)
Lag Revolving Rate	0.012	0.009	0.07
	(0.143)	(0.154)	(0.197)
Age	-0.008	-0.043	0.048
	(0.119)	(0.13)	(0.129)
Hq in Same Province	0.004	0.008	0.013
	(0.013)	(0.014)	(0.02)
Lag Leverage	-1.128	-1.497	-1.13
	(1.016)	(1.119)	(1.372)
Lag Ebitda/Assets	0.07	0.098	0.145
	(0.266)	(0.31)	(0.34)
Lag log(Assets)	0.025	0.031	0.024
	(0.05)	(0.045)	(0.045)
Multi. Rel.	-0.004	-0.005	-0.011
	(0.012)	(0.013)	(0.013)
Lag Liquidity	0.002	0.002	0.003
	(0.002)	(0.002)	(0.003)
Lag Investment	-0.001	-0.001	-0.001
	(0.002)	(0.003)	(0.004)
Risk Score	-0.085	-0.106	-0.094
	(0.059)	(0.082)	(0.097)
Lag CET1 Ratio	-0.0005	-0.001	-0.001
	(0.005)	(0.005)	(0.005)
Lag Liquidity	0.0007	-0.0009	-0.002
	(0.02)	(0.02)	(0.02)
Lag Retail Fund.	-0.004	-0.004	-0.003
	(0.076)	(0.077)	(0.075)
Lag Whole Fund.	-0.013	0.006	0.008
	(0.091)	(0.093)	(0.092)
Lag Bank Size	0.0002	-0.00002	0.0001
	(0.007)	(0.007)	(0.007)
BCC Dummy	-0.002	-0.011	-0.009
	(0.02)	(0.021)	(0.02)
Basel III CET1 Gap	0.001	0.001	0.002
	(0.01)	(0.011)	(0.011)

Table A1: Continuity of Covariates

Robust std. errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Note: The Table reports the statistical significance and coefficients' values for discontinuities in each of the covariates included in the covariates augmented version of Equation 3. This means the following specification: covariate_{bf} = $b_0 + b_1 R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + e_{bf}$, estimated locally, with a triangular kernel. Here x_{bf} is drawn credit, \bar{x} the \in 1.5 million threshold, ϕ the x_{bf} polynomial independently estimated on the two sides of the threshold, and the null hypothesis is $b_1 = 0$.

	RD, Pol(0)	RD, $Pol(1)$	RD, Pol(2)
	Firm Clustered Errors	Firm Clustered Errors	Firm Clustered Errors
$\hat{\beta} \ 2014$	-22.936^{***}	-22.644^{***}	-24.887^{***}
	(7.618)	(7.89)	(9.495)
Obs. (left; right)	2,853; 2,079	12,018; 5,112	23,874; 5,921
MSE Optimal Bdw.	305.73-389.93	753.16 - 2, 116.434	$991.995 \text{-} 4,\!685.959$
\hat{eta} 2013	2.261	1.923	-0.09
	(5.863)	(6.752)	(8.849)
Obs. (left; right)	5,410; 4,714	$17,675;\ 7,269$	16,482;7,905
MSE Optimal Bdw.	424.72-913.37	823.26-2,916.233	798.46-5,087.99
$\hat{\beta}$ 2014 (Non-SMEs)	-8.055	-6.681	18.706
	(14.119)	(17.177)	(29.268)
Obs. (left; right)	358; 2,536	766; 3,515	555; 3,842
MSE Optimal Bdw.	$297.152 ext{-} 3,\!544.12$	579.06 - 10,443.26	$454.09 - 21,\!636.88$
Controls	\checkmark	\checkmark	\checkmark

Table A2: **Dependent Variable** Interest Rate Change in *bp*; **Method** Simple RD

Robust standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Note: This Table presents the results of discontinuity tests run via local estimation of $\hat{\beta}$ in Equation 3: $\Delta i_{bf} = \alpha + \beta R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + \Omega C_{bf} + \nu_{bf}$, where Δi is the interest rate change in basis points, x_{bf} the past drawn credit, \bar{x} the $\in 1.5$ million threshold, ϕ the x_{bf} polynomial independently estimated on the two sides of the threshold, C_{bf} is a matrix of controls, and the null hypothesis of each test is $\beta = 0$. The different columns report increasing polynomial specifications. We compute estimates for the SMEs 2014 sample, the SMEs 2013 and non-SMEs 2014 placebo samples. We employ triangular kernel weights, and display robust standard errors clustered at the firm level in parentheses.

Controls: **Relationship level**: Lags of share of total drawn credit, revolving granted/total granted, utilized/granted, firm's and bank's hq in same province, log(relationship age). **Firm level**: Lags of liquidity ratio, leverage, log(assets), log risk score (Altman z-score), EBITDA/assets, industry dummies, regional dummies, dummy for the presence of multiple relationships, investment ratio. **Bank level**: Lags of tier 1 capital ratio, liquidity, retail funding ratio, wholesale funding ratio, log(assets), a BCC dummy.

	RD, $Pol(0)$	RD, Pol(1)	RD, Pol(2)
\hat{eta} 2014	-15.587^{***} (5.15)	-22.244^{***} (6.917)	-26.017^{***} (8.402)
Obs. (left; right)	4,816; 3,181	9,450; 5,675	17,948; 6,232
MSE-Optimal Bdw.	447.62 - 695.61	659.72 - 2,811.44	981.32 - 5,649.42
\hat{eta} 2013	3.25 (4.12)	1.505 (5.511)	5.599 (6.918)
Obs. (left; right)	10,326; 5,021	24,511; 7,469	37,293; 8,204
MSE-Optimal Bdw.	618.58 - 967.23	930.34 - 2,883.47	1,063.78 - 5,551.15
$\hat{\beta}$ 2014 (Non-SME)	-3.813 (13.044)	-7.851 (17.03)	1.204 (29.56)
Obs. (left; right)	328; 2,774	701; 3,825	686; 4,126
MSE-Optimal Bdw.	237.15 - 3,657.1	496.66 - 10,802.43	487.75 - 20,095.03
Standard errors in	parentheses		

Table A3: Dependent Variable Interest Rate Change in bp; No correction

*** p<0.01, ** p<0.05, * p<0.1

Note: This Table presents the results of discontinuity tests run *via* local estimation of $\hat{\beta}$ in Equation 3: $\Delta i_{bf} = \alpha + \beta R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + \nu_{bf}$, where Δi is the interest rate change in basis points, x_{bf} the past drawn credit, \bar{x} the \in 1.5 million threshold, ϕ the x_{bf} polynomial independently estimated on the two sides of the threshold, and the null hypothesis of each test is $\beta = 0$. Point estimates and errors are, in this case, not corrected for bandwidth selection. The different columns report increasing polynomial specifications. We compute estimates for the SMEs 2014 sample, the SMEs 2013 and non-SMEs 2014 placebo samples. We employ triangular kernel weights, and display robust standard errors clustered at the firm level in parentheses.

Bandwidth	25-25	35-35	45-45	55-55	65-65	75-75	85-85	95-95	105-105
^									
β 2014	-54.54^{**}	-48.093^{**}	-43.043^{**}	-38.195^{**}	-34.968^{**}	-32.736^{**}	-31.813^{**}	-31.368^{**}	-30.307^{**}
	(24.98)	(20.944)	(18.193)	(16.306)	(14.909)	(13.849)	(13.04)	(12.348)	(11.741)
Obs. (left; right)	184; 234	268; 301	352; 375	442; 446	528; 515	613; 571	690; 633	776;693	870; 762
$\hat{\beta}$ 2013	-4.187	-2.916	0.733	3.1	3.194	1.775	1.109	1.101	-0.755
	(20.473)	(17.423)	(15.468)	(14.118)	(13.017)	(12.163)	(11.479)	(10.911)	(10.424)
Obs. (left; right)	262; 296	340; 401	448; 487	556; 580	654; 684	739;765	836; 841	$935;\ 907$	$1,059;\ 981$

Table A4: Dependent Variable Interest Rate Change in bps; Hand-picked Bandwidth

Standard errors in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Note: This Table presents the results of discontinuity tests run via local estimation of $\hat{\beta}$ in a simplified version of Equation 3: $\Delta i_{bf} = \alpha + \beta R_{bf} + \nu_{bf}$, where Δi is the interest rate change in basis points, x_{bf} the past drawn credit, \bar{x} the $\in 1.5$ million threshold, and the null hypothesis of each test is $\beta = 0$. The different columns report increasingly larger hand-picked bandwidths for a linear polynomial specification of ϕ , with bandwidth size in thousands of Euros reported in the column header. We compute estimates for the SMEs 2014 sample, the SMEs 2013 and non-SMEs 2014 placebo samples. We employ triangular kernel weights, and display robust standard errors clustered at the firm level in parentheses.

	RD, $Pol(0)$	RD, $Pol(1)$	RD, $Pol(2)$
\hat{eta} 2014	0.002 (0.006)	0.006 (0.012)	0.014 (0.014)
Obs. (left; right)	$5,676;\ 6,327$	9,051; 9,796	11,997; 10,401
MSE-Optimal Bdw.	371.47 - 1,040.96	505.83 - 4,282.74	601.45 - 8,824.57
\hat{eta} 2013	-0.006 (0.009)	-0.006 (0.012)	-0.006 (0.013)
Obs. (left; right)	4,080; 9,881	8,096; 12,815	13,702; 13,526
MSE-Optimal Bdw.	250.05 - 1,678.1	423.35 - 4,844.43	581.7 - 9,343.82
$\hat{\beta}$ 2014 (Non-SME)	-0.03 (0.025)	-0.033 (0.032)	-0.038 (0.042)
Obs. (left; right)	570; 3,773	$1,152;\ 6,227$	1554; 6,682
MSE-Optimal Bdw.	248.22 - 2,624.72	487.92 - 13,292.23	594.88 - 28,113.90

Table A5: Dependent Variable Log changes in granted credit; Method Simple RD

Note: This Table presents the results of discontinuity tests run via local estimation of $\hat{\beta}$ in an alternative to Equation 3: $\Delta \log(\text{granted})_{bf} = \alpha + \beta R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + \nu_{bf}$, where $\Delta \log(\text{granted})$ is the log change in granted credit, x_{bf} the past drawn credit, \bar{x} the $\in 1.5$ million threshold, ϕ the x_{bf} polynomial independently estimated on the two sides of the threshold, and the null hypothesis of each test is $\beta = 0$. The different columns report increasing polynomial specifications. We compute estimates for the SMEs 2014 sample, the SMEs 2013 and non-SMEs 2014 placebo samples. We employ triangular kernel weights, and display robust standard errors clustered at the firm level in parentheses. We report observations left and right of the cutoff and the corresponding optimal bandwidths (in thousands of Euros) below each estimates' block.

	Dep. Variable:		Rates Ch	ange, bps			Granted,	Log Chan	ge
		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	SME-SF	-20.91^{***} (-3.10)	-19.58*** (-2.90)	-29.06*** (-3.87)	-28.44*** (-3.63)	$.0015 \\ (0.15)$.0024 (0.24)	$.0145 \\ (1.05)$.0174 (1.21)
Drawn/	Basel III Gap		-20.61*** (-4.82)		-20.63*** (-4.84)		0081^{***} (-2.76)		008^{***} (-2.69)
Granted Panel	Basel III Gap * SF		-5.062*** (-3.32)		-5.099*** (-3.26)		0029 (-1.08)		0032 (-1.22)
	Drawn/Granted			8.154 (1.17)	7.895 (1.11)			$.0215^{**}$ (2.28)	$.0219^{**}$ (2.29)
	Drawn/Granted * SF			(2.14)	12.77^{**} (2.41)			0179** (-2.09)	0207** (-2.32)
	SME-SF			8016 (-0.06)	6387 (-0.05)			0391** (-2.24)	0425** (-2.36)
Relationships	Basel III Gap				-20.7^{***} (-4.81)				008*** (-2.72)
Panel	Basel III Gap * SF				-4.863^{***} (-3.14)				0034 (-1.28)
	Multi Rel.			3.807 (0.39)	2.08 (0.19)			0089 (-0.91)	0101 (-1.01)
	Multi Rel. * SF			-21.16** (-2.09)	-19.94* (-1.81)			.0443*** (3.29)	$.0488^{***}$ (3.50)
	SME-SF			-23.41^{***} (-3.53)	-21.98*** (-3.26)			.0024 (0.24)	.0035 (0.34)
EBITDA	Basel III Gap				-20.43*** (-4.80)				0082*** (-2.8)
Panel	Basel III Gap * SF				-5.294*** (-3.41)				0027 (-0.04)
	EBITDA			-3.55 (-1.11)	-4.064 (-1.28)			$.0221^{***}$ (3.42)	$.0213^{***}$ (3.23)
	EBITDA * SF			-9.679** (-2.21)	-9.285** (-2.04)			.0049 (0.87)	.0061 (1.10)
	SME-SF			-25.97***	-25.28***			.0162	.0177
Risk	Basel III Gap			(-3.88)	(-3.74) -20.54*** (-4.82)			(1.46)	(1.54) 0083*** (-2.85)
Panel	Basel III Gap * SF				-5.886*** (-3.30)				0019 (-0.77)
	Log Risk Score			17.63 (1.58)	19.08^{*} (1.67)			.0041 (0.34)	.0049 (0.40)
	High Risk * SF			12.51^{***} (4.72)	14.23^{***} (5.57)			0426*** (-4.83)	0444*** (-4.94)
	Linear Rel., Firm, Bank Contro Observations	ols \checkmark	√ √ 4 14.367	√ √ 14.644	√ √ 14 367	√ √ 18 914	√ √ 0 17 796	√ √ 18 919	√ √ 17 796

Table A6: SME-SF's Effect Heterogeneity, Tier 1 Capital Robustness

Note: This Table explores how credit scarcity, proxied by different relationship or firm-level characteristics, and bank capital scarcity, proxied by the difference between bank's Tier 1 under the transitional and full Basel III phase-in definitions (Basel III Gap), interact with the SME-SF pass-through. The first four columns report effects on rates; first, the baseline effect of a linear local specification, for reference; second, the interaction between SME-SF Eligibility and bank capital scarcity; third, the interaction with credit scarcity; fourth, the model including both interaction terms. The second four columns report effects on granted credit. The four different panels use alternative proxies for credit scarcity. We estimate all specifications locally with triangular kernel weights, over bandwidths chosen to minimize MSE. t-Statistics, from errors clustered at the firm and bank level, are in parentheses. Controls: Relationship-level: Lags of share of total drawn credit, revolving granted/total granted, utilized/granted, firm's and bank's hq in same province, log(relationship age). Firm-level: Lags of liquidity ratio, leverage, log(assets), log risk score (Altman z-score), EBITDA/assets, industry dummies, regional dummies, dummy for the presence of multiple relationships, investment ratio. Bank-level: A dummy singling out cooperative banks; the lag of CET1, liquidity, retail and wholes ale funding ratios; the log of lag assets. $11\,$

	Dep. Variable:		Rates Ch	ange, bps			Granted,	Log Chan	ge
		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	SME-SF	-20.91*** (-3.10)	-18.91*** (-2.70)	-29.06*** (-3.87)	-27.79^{***} (-3.55)	.0015 (0.15)	.0023 (0.22)	.0145 (1.05)	.0168 (1.13)
Drawn/	Basel III Gap		-21.79*** (-7.10)		-21.77*** (-7.14)		0088*** (-2.65)		0088*** (-2.60)
Granted Panel	Basel III Gap * SF		-6.524*** (-2.87)		-6.623*** (-2.89)		.0011 (0.22)		.001 (0.21)
	Drawn/Granted			8.154 (1.17)	9.553 (1.31)			$.0215^{**}$ (2.28)	$.0234^{**}$ (2.38)
	Drawn/Granted * SF			11.73^{**} (2.14)	12.74^{**} (2.41)			0179** (-2.09)	0199** (-2.22)
	SME-SF			8016 (-0.06)	.9065 (0.07)			0391** (-2.24)	047^{**} (-2.55)
Relationships	Basel III Gap				-21.89*** (-7.14)				0087*** (-2.60)
Panel	Basel III Gap * SF				-6.312*** (-2.83)				6.9e-04 (-0.53)
	Multi Rel.			3.807 (0.39)	2.518 (0.22)			0088 (-0.91)	01 (-0.98)
	Multi Rel. * SF			-21.16** (-2.09)	-20.81^{*} (-1.74)			$.0443^{***}$ (3.29)	$.0535^{***}$ (3.90)
	SME-SF			-23.41*** (-3.53)	-21.67*** (-3.11)			.0024 (0.24)	.0037 (0.35)
EBITDA	Basel III Gap			. ,	-21.61*** (-7.10)			· /	0089*** (-2.67)
Panel	Basel III Gap * SF				-6.763^{***} (-2.91)				.0013 (0.26)
	EBITDA			-3.55 (-1.11)	-3.582 (-1.10)			$.0221^{***}$ (3.42)	.0216*** (3.27)
	EBITDA * SF			-9.819** (-2.24)	-10.91** (-2.47)			$.0049 \\ (0.88)$.0079 (1.42)
	SME-SF			-25.97***	-24.56***			.0162	.0186
Risk	Basel III Gap			(-3.88)	(-3.52) -21.79*** (-7.15)			(1.46)	(1.56) 0089*** (-2.63)
Panel	Basel III Gap * SF				-5.886*** (-3.30)				.0016 (0.36)
	Log Risk Score			17.63 (1.58)	18.95 (1.61)			.0041 (0.34)	.0042 (0.34)
	High Risk * SF			12.51^{***} (4.72)	14.08^{***} (5.70)			0426*** (-4.83)	0479*** (-5.12)
	Linear Rel., Firm, Bank Contro Observations	√ ols √ 14.64	√ √ 14 13.817	√ √ 7 14.644	√ √ 13.817	√ √ 18.21	$\begin{array}{c} \checkmark \\ \checkmark \\ 2 & 17.059 \end{array}$	√ √ 18.212	√ √ 17.059

Table A7: SME-SF's Effect Heterogeneity, 2013 Capital Gap Robustness

Note: This Table explores how credit scarcity, proxied by different relationship or firm-level characteristics, and bank capital scarcity, proxied by the difference between bank's CET1 as reported at December 2013 and full Basel III phase-in definitions (Basel III Gap), interact with the SME-SF pass-through. The first four columns report effects on rates; first, the baseline effect of a linear local specification, for reference; second, the interaction between SME-SF Eligibility and bank capital scarcity; third, the interaction with credit scarcity; fourth, the model including both interaction terms. The second four columns report effects on granted credit. The four different panels use alternative proxies for credit scarcity. We estimate all specifications locally with triangular kernel weights, over bandwidths chosen to minimize MSE. t-Statistics, from errors clustered at the firm and bank level, are in parentheses. *Controls*: **Relationship-level**: Lags of share of total drawn credit, revolving granted/total granted, utilized/granted, firm's and bank's hq in same province, log(relationship age). **Firm-level**: Lags of liquidity ratio, leverage, log(assets), log risk score (Altman z-score), EBITDA/assets, industry dummies, regional dummies, dummy for the presence of multiple relationships, investment ratio. **Bank-level**: A dummy singling out cooperative banks; the lag of CET1, liquidity, retail and wholesale funding ratios; the log of lag assets.

	WRD, Pol(1)	WRD, Pol(2)	WRD, $Pol(1)$	WRD, Pol(2)
	Double Clustered Errors	Double Clustered Errors	Double Clustered Errors	Double Clustered Errors
$\hat{\beta} 2014$	-35.374^{***}	-35.111^{***}	-38.648^{***}	-37.628^{***}
Eligible firms	(10.321)	(11.57)	(10.795)	(11.812)
Clusters	2,931 (Firms), 93 (Banks)	4,849 (Firms), 94 (Banks)	2,852 (Firms), 90 (Banks)	4,717 (Firms), 91 (Banks)
N. Observations	7,566	13,081	7,340	12,695
$\hat{\beta}$ 2013	7.034	2.506	6.807	1.56
Eligible firms	(8.441)	(10.594)	(7.941)	(10.806)
Clusters	6,484 (Firms), 96 (Banks)	9,576 (Firms), 96 (Banks)	6,313 (Firms), 96 (Banks)	9,329 (Firms), 96 (Banks)
N. Observations	17,368	26,346	16,935	25,690
$\hat{\beta} 2014$	-1.941	15.126	2.293	23.180
Non-Eligible firms	(20.360)	(32.773)	(21.234)	(35.1)
Clusters	1,047 (Firms), 77 (Banks)	1,089 (Firms), 78 (Banks)	1,013 (Firms), 75 (Banks)	1,051 (Firms), 76 (Banks)
N. Observations	4,006	4,285	3,847	4,112
Rel. Controls			\checkmark	\checkmark
Bank. Controls			\checkmark	\checkmark
Firm FE	\checkmark	\checkmark	\checkmark	\checkmark

Table A8: Dependent Variable Rate Change in bp; Method Fixed Effects RD

Note: This Table presents the results of discontinuity tests run via local estimation of $\hat{\beta}$ in Equation 3, augmented with fixed effects: $\Delta i_{bf} = \alpha + \beta R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + f + \nu_{bf}$, where Δi is the interest rate change in basis points, x_{bf} the past drawn credit, \bar{x} the $\in 1.5$ million threshold, ϕ the x_{bf} polynomial independently estimated on the two sides of the threshold, f firm fixed effect, and the null hypothesis of each test is $\beta = 0$. The different columns report increasing polynomial specifications, and - final columns - the estimates of the linear polynomial specification adjusted for covariates insertion. We compute estimates for the SMEs 2014 sample, the SMEs 2013 and non-SMEs 2014 placebo samples. We employ triangular kernel weights, and display in parentheses robust standard errors, double-clustered at the bank and firm level. The acronym WRD stands for "within RD".

Controls: **Relationship level**: Lags of share of total drawn credit, revolving granted/total granted, utilized/granted, firm's and bank's hq in same province, log(relationship age). **Firm level**: Lags of liquidity ratio, leverage, log(assets), log risk score (Altman z-score), EBITDA/assets, industry dummies, regional dummies, dummy for the presence of multiple relationships, investment ratio. **Bank level**: Lags of tier 1 capital ratio, liquidity, retail funding ratio, wholesale funding ratio, log(assets), a BCC dummy.

Simple RDD on Firm-FE Sample					
$\hat{\beta}$ 2014 Linear	-41.39***	-39.60***	-45.97***		
	(-3.94)	(-4.31)	(-4.84)		
Observations	$5,\!087$	5,036	$4,\!959$		
$\hat{\beta}$ 2014 Quadratic	-43.50***	-38.12***	-43.61***		
	(-4.29)	(-3.87)	(-4.76)		
Observations	$7,\!686$	$5,\!842$	5,753		
Non-Eligible Lines, Firm-FE vs Full Samples					
$\hat{\gamma}$ 2014 Linear	2.011	0.358	-1.902		
	(0.36)	(0.06)	(-0.29)		
Observations	$5,\!654$	$5,\!580$	$5,\!478$		
$\hat{\gamma}$ 2014 Quadratic	2.672	-0.118	-2.988		
	(0.52)	(-0.02)	(-0.42)		
Observations	6,202	$6,\!117$	6,007		
Firm Controls	\checkmark	\checkmark	\checkmark		
Bank Controls		\checkmark	\checkmark		
Relationship Controls			\checkmark		

Table A9: **Dependent Variable** Interest Rate Change in *bp*; FE-Sample estimates

t statistics in parentheses

*** p<0.01, ** p<0.05, * p<0.1

Note: This Table presents, in the first two panels, the results of discontinuity tests run via local estimation of $\hat{\beta}$ in Equation 3: $\Delta i_{bf} = \alpha + \beta R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}|) + \nu_{bf}$, on the sub-sample of observations on which the fixed effect estimator of β is identified. Such sub-sample consists of all the observations belonging to firms that have at least one eligible and one non-eligible observation within the bandwidth selected by minimizing the MSE. Δi is the interest rate change in basis points, x_{bf} the past drawn credit, \bar{x} the \in 1.5 million threshold, ϕ the x_{bf} polynomial independently estimated on the two sides of the threshold, and the null hypothesis of each test is $\beta = 0$. The different columns report specification including different controls. In the second two panels, the Table reports the results of the comparison between the pre-post reform change in the rates of non-eligible relationships within the firm fixed effects sample, and within the overall sample. Each time, we select the relationships within the right side of the data-driven bandwidth of the respective (1st or 2nd order) specification, and run the following test: $\Delta i_{bf} = \eta + \gamma S_{bf} + \phi_+ (|x_{bf}^{2013} - \bar{x}|) + \Omega C_{bf} + \epsilon_{bf}$. Δi_{bf} is the interest rate change in basis points, S_{bf} is a dummy equal to one if the observation falls in the local sub-sample for which the firm fixed effect is identified, x_{bf} the past drawn credit, \bar{x} the $\in 1.5$ million threshold, ϕ_+ the right x_{bf} polynomial independently estimated on the two sides of the threshold, C_{bf} includes other covariates, ν_{bf} is the stochastic error term, for which we allow clustering at the bank and firm level, and the null hypothesis of each test is $\gamma = 0$. The different columns report specifications including different controls.

Controls: **Relationship level**: Lags of share of total drawn credit, revolving granted/total granted, utilized/granted, firm's and bank's hq in same province, log(relationship age). **Firm level**: Lags of liquidity ratio, leverage, log(assets), log risk score (Altman z-score), EBITDA/assets, industry dummies, regional dummies, dummy for the presence of multiple relationships, investment ratio. **Bank level**: Lags of tier 1 capital ratio, liquidity, retail funding ratio, wholesale funding ratio, log(assets), a BCC dummy.



Figure A1: Permutation Test

Note: The Figure reports the point estimates and confidence intervals (gray shaded areas) for discontinuities in the cost of credit dynamics at alternative thresholds, progressively more distant from the true SME-SF assignment cut-off. Black diamonds represent point estimates obtained at the displaced thresholds, while the red square marks the estimate at the real SME-SF assignment threshold. We estimate the following specification $\Delta i_{bf} = a + \beta R_{bf} + \phi(|x_{bf}^{2013} - \bar{x}_p|) + \nu_{bf}$ over MSE-minimizing bandwidths, different at the two sides of the threshold, and employing a triangular kernel. Here x_{bf} is past drawn credit in thousand of Euros, \bar{x}_p the assignment threshold we choose, ϕ the linear x_{bf} polynomial independently estimated on the two sides. Below the plot, under the label "Threshold", each line marks for which \bar{x}_p we estimated the model. A black dot to the right of the corresponding line marks that the estimate above and its confidence interval concern that specific threshold \bar{x}_p .